

Temi di Discussione

(Working Papers)

A quantitative analysis of risk premia in the corporate bond market

by Sara Cecchetti

bag 1141



Temi di discussione

(Working papers)

A quantitative analysis of risk premia in the corporate bond market

by Sara Cecchetti

Number 1141 - October 2017

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Ines Buono, Marco Casiraghi, Valentina Aprigliano, Nicola Branzoli, Francesco Caprioli, Emanuele Ciani, Vincenzo Cuciniello, Davide Delle Monache, Giuseppe Ilardi, Andrea Linarello, Juho Taneli Makinen, Valerio Nispi Landi, Lucia Paola Maria Rizzica, Massimiliano Stacchini. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

A QUANTITATIVE ANALYSIS OF RISK PREMIA IN THE CORPORATE BOND MARKET

by Sara Cecchetti*

Abstract

We propose an econometric model to decompose corporate bond spreads into compensation required by investors for unpredictable future changes in the credit environment and for expected default losses. We use the model to understand whether the significant reduction in corporate bond spreads observed since the launch of the CSPP (Corporate Sector Purchase Programme) is attributable more to the fact that expansionary monetary policy measures tend to increase the risk appetite of investors and compress risk premia, or to the ability of unconventional measures to reduce expected default losses by improving investors' expectations about the economic and financial conditions of issuers.

JEL Classification: B26, C02, F30, G12, G15.

Keywords: bond excess return, credit default swap, distress risk premium, expected default frequency, jump-at-default risk premium.

Contents

1. Introduction	5
2. Two components of corporate bonds excess return: the model	8
2.1 Price of a Credit Default Swap	9
2.2 Distress risk premium	11
2.3 Jump-at-default risk premium	14
3. Data and econometric framework	15
3.1 Maximum likelihood estimates of risk-neutral default intensity	
3.2 Maximum likelihood estimates of actual default intensity	
4. Estimates of the two risk premia components	19
4.1 Estimation results of the distress risk premium	19
4.2 Estimation results of the jump-at-default risk premium	
5. Analysis of the two risk premium components and event study exercise	
on the risk premia and the expected losses	
6. Conclusion	
Appendix	
References	

^{*} Bank of Italy, Directorate General for Economics, Statistics and Research.

1 Introduction¹

Corporate bond spreads are carefully monitored by central banks as they influence the transmission of the monetary policy decisions to the real economy, thus determining their effectiveness. From a monetary policy perspective, the aim of this paper is to evaluate the effectiveness of the Corporate Sector Purchase Programme (CSPP) launched by the ECB in March 2016 in the corporate bond market. In particular, the main objective of our analysis is to understand if the significant reduction in corporate bond spreads observed since the launch of the CSPP can be attributed more to the ability of unconventional measures to reduce expected losses by improving investors' expectations about the economic and financial conditions of issuers or to the fact that expansionary monetary policy measures tend to increase the risk appetite of investors and consequently compress risk premia. It is well known that these two different aspects are possible and desirable effects of unconventional measures and both are reflected in corporate bond spreads as the compensation for the expected losses and the bond risk premium required by investors, respectively.²

Instead of using corporate bond spreads, we rely on the information embedded in CDS prices. In fact, default swap spreads approximate the spreads of referenced bonds.³ Moreover, empirical literature⁴ suggests that CDS spreads are better measures of default risk than bond spreads for a number of reasons. Roughly speaking, mainly because: the corporate CDS market is more liquid than the corresponding bond market; it is difficult to construct bond spreads in practice (for example in terms of maturity matching); it is well known that the CDS market plays a leading role in the price discovery process.⁵

We decompose the price of a credit default swap into an expected losses component and a risk premium component, which we then further decompose into two contributions.

¹I would like to thank Lorenzo Braccini, Francesco Corsello, Davide Delle Monache, Antonio Di Cesare, Stefano Neri, Pedro Serrano, Marco Taboga and Gabriele Zinna for their helpful comments and suggestions. The views expressed in this paper are those of the author and do not necessarily reflect those of the Bank of Italy. All the remaining errors are my own. E-mail: sara.cecchetti@bancaditalia.it

²For a discussion of the decomposition of corporate bond spreads in the expected losses and the risk premium components see, for example, Section 2.2 in Cecchetti and Taboga (2017).

³In Berndt and Obreja (2010) the payments of a CDS can be reproduced by a replicating portfolio consisting of a long position in a defaultable bond and a short position in a risk-free bond.

⁴See Blanco et al. (2005) and Forte and Pena (2009).

 $^{^{5}}$ Meaning that price variations of CDSs anticipate variations in bond spreads. This evidence is consistent with the standard hypothesis whereby CDS prices adjust more rapidly to the release of new information and that adjustment, in turn, generates an informative signal to which bond spreads react with a time lag.

In particular, building on the work of Díaz et al. (2013), we propose econometric models to decompose risk premia into two different sources of risk:⁶ 1) the compensation for changes in the credit environment associated with business and macro conditions (and thus for unexpected changes in the creditworthiness of the bond issuer), which investors receive for bearing risks associated with unpredictable variations in the underlying state variables; 2) the remuneration required for the risk associated with the restructuring of the entity (and thus for the risk that the bond's price will drop in the event of default). Disentangling these two compensations allows us to quantify the different contributions of the sources of the excess return in a corporate bond.

We monitor the evolution over time of the estimated expected losses and two risk premia components and we focus on the most recent period to assess on which of them the CSPP has been more effective.

To estimate the expected losses component we only need CDS prices: in fact, this component is just approximated by the CDS price under the objective probability measure. To estimate the two abovementioned risk premium components we need both CDS prices and expected default frequency (EDF) data: in particular, we employ the only information embedded in CDS contracts by focusing on the default risk premium embedded in such spreads to quantify the compensation for those future changes in the creditworthiness of the bond issuer that might vary from expectations,⁷ the so-called *distress risk premium*; to quantify the remuneration for the surprise jump in the bond price at the event of default - the *jump-at-default-risk premium* - we also need EDF data, used as a proxy of the actual probability of default of an issuer.

From a methodological point of view, this article adopts the intensity approach of Lando (1998) and Duffie and Singleton (1999) to study the default risk premium. Our estimation strategy follows a two-step procedure, as initially developed by Driessen (2005) and employed by Díaz et al. (2013).

First, we obtain the maximum likelihood (ML) estimates of the risk-neutral mean of default arrival rates ($\lambda^{\mathbb{Q}}$). To do this we use a dataset of source Capital IQ composed of the daily spreads of 1-, 3- and 5-year CDS contracts for the firms in the iTraxx Europe index, from January 2007 to February 2017. Under the assumption that the default event is diversifiable, we can obtain the estimates of the distress risk premium for each

⁶The same decomposition is employed by Pan and Singleton (2006) for the credit spreads of Japanese banks.

 $^{^7\}mathrm{As}$ in Pan and Singleton (2008) for sovereign CDS spreads.

firm. The same procedure was previously implemented by Pan and Singleton (2008) and Longstaff et al. (2011) in the case of sovereign CDS spreads, and by Díaz et al. (2013) for corporate CDS spreads.⁸ In this first step we also extrapolate the informations about the expected losses embedded in CDS prices by looking at the CDS price in terms of the risk neutral default intensity but under the objective probability measure \mathbb{P} (thus not reflecting investors' risk preferences).

Second, we estimate the ML actual default intensity $(\lambda^{\mathbb{P}})$ using the information at our disposal on the actual probabilities of default as those provided by Moody's KMV expected default frequencies (EDFs). The jump-at-default risk premium for each firm is computed in terms of the ratio $\lambda^{\mathbb{Q}}/\lambda^{\mathbb{P}}$, as shown in Yu (2002), Driessen (2005) and Pan and Singleton (2006).

In this way we assess fully both the expected losses and the corporate default risk premium embedded in European CDS of each firm in the iTraxx index.

The two components of risk premium and the expected losses component are investigated to determine the main channel by which the CSPP has determined a significant reduction in corporate bond spreads. For this reason we conduct an event study analysis around the main dates of the announcements of the ECB related to the CSPP. We can confirm that the date on which both components of risk premium and the expected losses component decreased the most is 10 March 2016, when the ECB announced its decision to launch the programme. Looking at the variations between the day before and the day after the announcement, we find a more important effect of the CSPP on the risk premia component and, in particular, on the distress risk premium; however, the reduction of the expected losses is also not negligible. Moreover, using our firm-by-firm analysis, we can investigate the different behaviour of the risk premium and expected losses components in the financial and non-financial sectors, and we are able to assess the spillovers of the CSPP⁹ to financial firms (as the iTraxx Europe index also includes 25 CDSs of financial firms): one interesting result of our paper is that there are, in fact, spillovers to the financial sector, in particular in terms of a reduction of expected losses. Considering the entire time period between the first announcement of CSPP and

⁸For the dynamics of $\lambda^{\mathbb{Q}}$, we also studied a different model (starting from Li and Zinna (2014) and Cheridito et al. (2007)) from the one used in those papers, but we found that this new model did not perform as well as the first: for this reason we continued our analysis using the consolidated model.

⁹In the CSPP the object of the purchases are the investment grade bonds with residual life between 1 and 30 years, issued by non-financial firms or non-banking financial firms. The purchases can be made both in the primary and the secondary markets.

February 2017, we can say that both expected losses and risk premia components have continued to diminuish, in both the financial and non-financial sectors.

We contribute to the literature by developing an analysis of the sources of both expected losses and sources of excess returns of corporate bonds (also known in the literature as *excess bond premium*¹⁰) during a time span that encompasses both periods of crisis (financial and sovereign debt) in the last ten years and the recent unconventional monetary policy measures implemented by the ECB. Our firm-by-firm analysis potentially allows different investigations, such as of differences between sectors of the firms or between different credit worthiness scores. From a monetary policy perspective, we are the first, to our knowledge, to use the consolidated framework for the CDS prices to assess the channels by which the recently launched CSPP has been effective in reducing corporate bond spreads. A significant contribution would derive from a comparison of the results obtained calibrating the model to the iTraxx index with those of a theoretical iTraxx index obtained from the aggregation of individual CDSs, in order to better disentangle systemic and idiosyncratic risks and contribute to the systemic risk literature; this will be subject of future research.

The paper is organized as follows. Section 2 presents the theoretical framework of the model used to decompose the sources of default risk premium of a general defaultable security and to identify the risk premia and expected losses components in a CDS price. Section 3 describes the data and the econometric framework for the estimation strategy. Section 4 contains the estimation results for the two components of the risk premium. Section 5 describes the event study analysis. Section 6 concludes. Proofs and technicalities are collected in the Appendix.

2 Two components of corporate bonds excess return: the model

Consider a defaultable security with price $P(t, X_t)$ depending on a set of state variables X_t that follow a diffusion process under a probability measure \mathbb{P}

$$dX_t = \mu_X^{\mathbb{P}}(X_t, t)dt + \sigma_X(X_t, t)dW_t^{\mathbb{P}},$$

where $\mu_X^{\mathbb{P}}$ and σ_X are the drift and the instantaneous volatility, respectively, and $W_t^{\mathbb{P}}$ is a standard Brownian motion under the actual measure \mathbb{P} . According to Girsanov's

 $^{^{10}}$ A completely different approach that also analysis the excess bond premium is in Gilchrist and Zakrajsek (2012).

theorem, let define a price of risk Λ_t and a Brownian motion under the risk-neutral measure \mathbb{Q}

$$W_t^{\mathbb{Q}} = W_t^{\mathbb{P}} + \int_0^t \Lambda_s ds$$

so that the risk-neutral process for the state variables becomes

$$dX_t = \mu_X^{\mathbb{Q}}(X_t, t)dt + \sigma_X(X_t, t)dW_t^{\mathbb{Q}},$$

with the drift under the risk neutral measure being $\mu_X^{\mathbb{Q}} = \mu_X^{\mathbb{P}}(X_t, t) - \sigma_X(X_t, t)\Lambda_t$.

The excess return (or risk premium) of the defaultable security can be written as the difference of expectations of the relative price variation under \mathbb{P} and \mathbb{Q} measures, and computed by considering $P(t, X_t)$ as a function of the state variables X_t and using standard techniques of stochastic calculus (basically apply Ito's lemma for stochastic processes with drift, diffusion and jumps, under \mathbb{Q} and \mathbb{P} measures):

$$e_t = \mathbb{E}^{\mathbb{P}}\left[\frac{dP(X_t, t)}{P(X_t, t)}\right] - \mathbb{E}^{\mathbb{Q}}\left[\frac{dP(X_t, t)}{P(X_t, t)}\right] = \frac{1}{P_t}\frac{\partial dP_t}{\partial X_t}\sigma_X(X_t, t)\Lambda_t + \frac{R_t - P_t}{P_t}\lambda_t^{\mathbb{P}}\Gamma_t$$
(1)

where Λ_t represents the market price of risk (risk premium per unity of volatility), $\sigma_X(X_t, t)$ is the volatility of risk factors, R_t is the recovery value, $\lambda_t^{\mathbb{P}}$ is the default intensity under the objective or risk-adjusted measure (or the arrival rate of a credit event), $\Gamma_t = 1 - \frac{\lambda^{\mathbb{Q}}}{\lambda^{\mathbb{P}}}$ is the price of risk at the event of default ($\lambda_t^{\mathbb{Q}}$ being the default intensity under the risk-neutral measure). The first term of equation 1 (price of risk mutiplied by volatility of risk factors) accounts for changes in the risk environment, and is called *distress risk premium*; the second term represents the expected payoff associated with a (downward) jump in the price of the bond if the reference entity does restructure, and is called *jump-at-default-premium*. A sketch of the proof of equation (1) can be found in Appendix A.1.

We can see that the excess return in equation (1) is zero if there is no compensation for both distress and jump-at-default risks ($\Lambda_t = \Gamma_t = 0$).

2.1 Price of a Credit Default Swap

A Credit Default Swap (CDS) is a financial swap agreement between two parties (protection seller and protection buyer) to receive insurance against the default of a certain bond (the reference entity). In the event of default the protection seller receives the defaulted bond, and restores its amount to the protection buyer. To get such insurance, the protection buyer pays a spread to the protection seller, usually quarterly up to the maturity of the contract, if there is no default.

Longstaff et al. (2005) and Pan and Singleton (2008) provide the following formula for the price of a CDS contract with maturity M in an intensity based setting:

$$CDS_t^{\mathbb{Q}}(M) = \frac{(1 - R_t^{\mathbb{Q}})\int_t^{t+M} \mathbb{E}_t^{\mathbb{Q}} [\lambda_u^{\mathbb{Q}} e^{-\int_t^u (r_s + \lambda_s^{\mathbb{Q}}) ds}] du}{\int_t^{t+M} \mathbb{E}_t^{\mathbb{Q}} [e^{-\int_t^u (r_s + \lambda_s^{\mathbb{Q}}) ds}] du}$$
(2)

where r_t is the risk-free interest rate, $\lambda_t^{\mathbb{Q}_{11}}$ and $(1 - R_t^{\mathbb{Q}})$ are default intensity and the loss given default under the risk-neutral \mathbb{Q} measure at time t. As in Pan and Singleton (2008), we assume independence between r_t and λ_t , and a constant recovery rate R.¹² Additionally we consider that the annual spread is usually paid quarterly. It follows that we can rewrite equation (2) as

$$CDS_t^{\mathbb{Q}}(M) = 4 * \frac{(1 - R^{\mathbb{Q}}) \int_t^{t+M} D(t, u) \mathbb{E}_t^{\mathbb{Q}} [\lambda_u^{\mathbb{Q}} e^{-\int_t^u \lambda_s^{\mathbb{Q}} ds}] du}{\sum_{i=1}^{4M} D(t, t+0.25i) \mathbb{E}_t^{\mathbb{Q}} [e^{-\int_t^{t+0.25i} \lambda_s^{\mathbb{Q}} ds}] du}$$
(3)

where $\mathbb{E}_t^{\mathbb{Q}}$ denotes expectations based on $\lambda_t^{\mathbb{Q}}$ following a risk-neutral stochastic process and D(t, u) is the price of a default-free zero-coupon bond (issued at date t and maturing at date u).

When we ask for the market price of a CDS, we refer to the price under the risk neutral measure \mathbb{Q} . With simple algebra, we can write this observed price in terms of the price under the objective measure \mathbb{P} as

$$CDS^{\mathbb{Q}} = CDS^{\mathbb{P}} + (CDS^{\mathbb{Q}} - CDS^{\mathbb{P}}), \tag{4}$$

where $CDS^{\mathbb{P}}$ represents the price component related to the objective probability of default, while $(CDS^{\mathbb{Q}}-CDS^{\mathbb{P}})$ represents a risk premium component. The price of a credit default swap can thus be decomposed into an expected losses component, approximated by the CDS price under the objective probability measure, and a risk premium component. In the following sections, we are going to decompose the risk premium component into the different contributions of the distress risk premium and the jump-at-default risk premium components.

¹¹Discounting by $r_t + \lambda_t^{\mathbb{Q}}$ captures the survival-dependent nature of the payments.

¹²These strong technical assumptions are standard in the litarature, and necessary to construct the discrete approximation for the solution of the CDS price introduced in Lando (2004), which we'll see in Section 2.2.

2.2 Distress risk premium

In formula (1) we have seen that the excess return, or risk premium component of the price of a defaultable security, can be decomposed into a distress risk premium component and a jump-at-default risk component. In the first step we assume no compensation for the default event ($\Gamma_t = 0$), and thus a null jump-at-default risk. From a theoretical point of view we are assuming the conditionally diversifiable hypothesis of Jarrow et al. (2005): this assumption states that jump-at-default risk is purely idiosyncratic when risk-neutral and actual default probabilities are equal, conditional to the existence of an infinite number of bonds in the economy and independence between default processes. Considering the high number of bonds in the iTraxx and the very low probability of a simultaneous default given the investment grade ratings, these two conditions seem to be reasonably satisfied.

To estimate the distress risk premium, we rely on the information embedded in CDS prices. In fact default swap spreads approximate the spreads of referenced bonds and empirical literature suggests to use CDS spreads to measure default risk.¹³

As in Díaz et al. (2013) we impose a Ornstein-Uhlenbeck process for the logarithm of the default intensity $\lambda_t^{\mathbb{Q}}$ under the risk-neutral measure \mathbb{Q} :

$$d\ln\lambda_t^{\mathbb{Q}} = K^{\mathbb{Q}}(\theta^{\mathbb{Q}} - \ln\lambda_t^{\mathbb{Q}})dt + \sigma_\lambda dW_t^{\mathbb{Q}}$$
(5)

where parameters $K^{\mathbb{Q}}$, $\theta^{\mathbb{Q}}$ and σ_{λ} capture the mean-reversion rate, the long-run mean and the volatility of the process, respectively. By adopting this framework, the intensity is ensured to be positive. In order to deal with the same stochastic process under the objective (or historical) measure \mathbb{P} , we assume a market price of risk Λ_t underlying a change of measure from \mathbb{P} to \mathbb{Q} to be an affine function of $\ln \lambda_t$:

$$\Lambda_t = \gamma_0 + \gamma_1 \ln \lambda_t. \tag{6}$$

In fact, the dynamics of the logarithm of the risk-neutral mean arrival rate of default $\lambda_t^{\mathbb{Q}}$ under the objective measure \mathbb{P} results in¹⁴

$$d\ln\lambda_t^{\mathbb{Q}} = K^{\mathbb{P}}(\theta^{\mathbb{P}} - \ln\lambda_t^{\mathbb{Q}})dt + \sigma_\lambda dW_t^{\mathbb{P}},\tag{7}$$

 $^{^{13}\}mathrm{See}$ the Introduction for references and a brief explanation.

¹⁴Technical proof can be found in Appendix A.3.

where the mean-reversion rate and the long-run mean of the process under the objective probability measure, in terms of the market price of risk parameters, are

$$K^{\mathbb{P}} = K^{\mathbb{Q}} - \gamma_1 \sigma_\lambda$$

and

$$K^{\mathbb{P}}\theta^{\mathbb{P}} = K^{\mathbb{Q}}\theta^{\mathbb{Q}} + \gamma_0\sigma_\lambda.$$

Note that we are discussing the properties of $\lambda_t^{\mathbb{Q}}$, as a stochastic process, under two different measures, \mathbb{Q} and \mathbb{P} . At this juncture, $\lambda_t^{\mathbb{P}}$, the arrival rate of default under the historical measure, is playing no role in our analysis. As highlighted by Jarrow et al. (2005) and Yu (2002), this information cannot be extracted from bond or CDS spread data alone. We will deal with $\lambda_t^{\mathbb{P}}$ and we will briefly comment on the relationship between $\lambda_t^{\mathbb{P}}$ and $\lambda_t^{\mathbb{Q}}$ in the subsequent section, when we will estimate the jump-at-default risk premium.

In our analysis we have also considered the CIR (square root) process for λ_t , as used in Li and Zinna (2014):

$$d\lambda_t^{\mathbb{Q}} = K^{\mathbb{Q}}(\theta^{\mathbb{Q}} - \lambda_t^{\mathbb{Q}})dt + \sigma_\lambda \sqrt{\lambda_t^{\mathbb{Q}}}dW_t^{\mathbb{Q}}$$
(8)

Using a CIR process for λ_t , has the advantage that both the expectations in the numerator and denominator of equation (3) can be computed in closed form (see Appendix A.5). Differently from Li and Zinna (2014), who adopt an essentially affine market price of risk as in Duffee (2002),¹⁵ we have assumed an extended market price of risk Λ_t from \mathbb{P} to \mathbb{Q} as in Cheridito et al. (2007):

$$\Lambda_t = \frac{\gamma_0}{\sqrt{\lambda_t}} + \gamma_1 \sqrt{\lambda_t}.$$

The advantage of this choice of market price of risk is that again the dynamics for the risk neutral intensity $\lambda_t^{\mathbb{Q}}$ under the actual measure \mathbb{P} is a CIR process ¹⁶

$$d\lambda_t^{\mathbb{Q}} = K^{\mathbb{P}}(\theta^{\mathbb{P}} - \lambda_t^{\mathbb{Q}})dt + \sigma_\lambda \sqrt{\lambda_t^{\mathbb{Q}}} dW_t^{\mathbb{P}}$$
⁽⁹⁾

with

$$K^{\mathbb{P}} = K^{\mathbb{Q}} - \sigma_{\lambda} \gamma_{1}$$

¹⁵According to the authors, the motivation for this choice is to avoid to impose the Feller condition required by the CIR dynamcis to assure a $\lambda_t^{\mathbb{Q}}$ strictly positive: $2K^{\mathbb{Q}}\theta^{\mathbb{Q}} > \sigma_{\lambda}^2$. Such choice of market price of risk determines only one different parameter in the processes under the actual and the risk neutral measure.

¹⁶Technical proof can be found in Appendix A.4.

$$K^{\mathbb{P}}\theta^{\mathbb{P}} = K^{\mathbb{Q}}\theta^{\mathbb{Q}} + \sigma_{\lambda}\gamma_0$$

so that we let vary both the speed of adjustment and the long run parameters of the process for the default intensity under the two measures \mathbb{Q} and \mathbb{P} .

The model based on the CIR dynamics however, empirically provides a worst performance when calibrated to the data, with respect to the model based on the Ornstein-Uhlenbeck process. For this reason, after looking at the results, we have decided to rely on the Ornstein-Uhlenbeck process, and in the following we will refer to this model.

To measure the size of the distress risk premium, we follow Longstaff et al. (2011): as the dynamics of the objective (under \mathbb{P} measure) and risk-neutral (under \mathbb{Q}) processes for $\lambda_t^{\mathbb{Q}}$ coincide when there is no risk premium ($\Lambda_t = 0$) since, from the above discussion, they would have the same parameters, the size of the risk premium can be inferred by simply taking the difference

$$DRP_t = CDS_t^{\mathbb{Q}}(M) - CDS_t^{\mathbb{P}}(M)$$
(10)

where $CDS_t^{\mathbb{Q}}(M)$ is the price of the CDS implied by the risk neutral process $\lambda_t^{\mathbb{Q}}$ (taking expectations in equation (3) using the risk-neutral probability distribution \mathbb{Q} implied by equation (5)), $CDS_t^{\mathbb{P}}(M)$ is the price of the CDS implied by the objective process (taking expectations in equation (3) but using the probability distribution \mathbb{P} implied by the objective process in equation (7)). Let us remark that the CDS price under the objective measure $CDS_t^{\mathbb{P}}(M)$ provides an estimate of the expected losses.

For the CDS price computation we use the general approximation formula described in Lando (2004):

$$CDS_t^{\mathbb{Q}}(M) = 4 * \frac{(1-R)\sum_{i=1}^M D(0,i)(S(0,i-1) - S(0,i))}{\sum_{i=1}^M D(0,i)S(0,i)}$$
(11)

where

$$S(0,t) = \mathbb{E}_0^{\mathbb{Q}} \left[e^{-\int_0^t \lambda_s^{\mathbb{Q}} ds} \right]$$

and we compute this expectation numerically using the Crank–Nicholson implicit finite-difference method to solve the associated Feynman-Kac partial differential equation:¹⁷

$$\begin{cases} S_t + S_x (K^Q \theta^Q - K^Q \ln \lambda_t) + \frac{1}{2} S_{xx} \sigma^2 - \lambda_t S = 0\\ S(t) = 1 \end{cases}$$

and

¹⁷See Appendix A.6 for details of this method.

2.3 Jump-at-default risk premium

Separate from the risk related to unexpected risk-neutral default arrivals because of changes in the credit environment (which we called *distress risk premium*), is the risk of a jump in the price of the underlying bond in the event that a firm restructures. We refer to this risk as the *jump-at-default risk*, for which investors also demand a *default risk premium*.

This compensation, as reflected in excess returns, is captured in the second term of equation (1), as

$$JADRP_t = \frac{R_t - P_t}{P_t} \lambda_t^{\mathbb{P}} \Gamma_t$$

where $\frac{R_t - P_t}{P_t}$ is the percentage loss of value due to default, $\lambda_t^{\mathbb{P}}$ is the actual default intensity and Γ_t is the market price of jump-at-default risk.

It turns out theoretically¹⁸ that the risk premium associated with the jump-at-default risk is the ratio between the risk-neutral and historical arrival rate of credit events $\lambda_t^{\mathbb{Q}}/\lambda_t^{\mathbb{P}}$, and the price of this risk can be defined as¹⁹

$$\Gamma^{\mathbb{P}}_t = 1 - \frac{\lambda^{\mathbb{Q}}_t}{\lambda^{\mathbb{P}}_t}$$

The ratio $\lambda_t^{\mathbb{Q}}/\lambda_t^{\mathbb{P}}$ is usually referred to as the jump-at-default premium and has been previously studied in Driessen (2005) and Berndt et al. (2005).

Typically, the ratio $\frac{\lambda_t^{\varphi}}{\lambda_t^{\mathbb{P}}}$ is larger than one since, assuming that investors are averse to jump-at-default risk, to obtain the correct market prices using risk-neutral valuation, $\lambda^{\mathbb{Q}}$ must be set larger than $\lambda^{\mathbb{P}}$. Effectively, the investment environment must, riskneutrally, be much riskier (default must be more likely) than what has been experienced historically.

If the jump-at-default premium is one (risk-neutral and actual default intensities coincide), then JADRP is zero and does not affect excess returns: in other words, there is no concern about jumps at the time of credit events and no compensation for the event of default. On the other hand, if the jump-at-default premium is greater than one, then Γ_t is negative. Since the term $\frac{R_t - P_t}{P_t}$ is negative if prices jump down when there is a credit event, the overall contribution of JADRP to the excess return is positive.

 $^{^{18}\}mathrm{See}$ Yu (2002) for a heuristic discussion.

 $^{^{19}\}mathrm{See}$ Appendix A.7 for a technical explanation.

While $\lambda_t^{\mathbb{Q}}$ (and the distress risk premium component) are obtained using only CDS data, according to general literature,²⁰ to estimate $\lambda_t^{\mathbb{P}}$ we rely on the market's expected probabilities of default over the next year, or Expected Default Frequencies (EDFs), calculated by Moody's KMV using a Merton-style balance-sheet model of credit events, as proxies for actual default probabilities.

More specifically, we assume that the actual default intensity process $\lambda_t^{\mathbb{P}}$ comes from the definition of the EDF over M years

$$EDF(M) = 1 - \mathbf{E}_t^{\mathbb{P}} \left[e^{-\int_t^{t+M} \lambda_s^{\mathbb{P}} ds} \right]$$
(12)

and, as for the default intensity under the risk neutral measure, we assume an Ornstein-Uhlenbeck process for the logarithm of the objective default intensity:

$$d\ln\lambda_t^{\mathbb{P}} = \alpha^{\mathbb{P}}(\beta^{\mathbb{P}} - \ln\lambda_t^{\mathbb{P}})dt + \sigma_{\lambda^{\mathbb{P}}}dW_t^{\mathbb{P}}$$
(13)

where parameters $\alpha^{\mathbb{P}}$, $\beta^{\mathbb{P}}$ and $\sigma_{\lambda^{\mathbb{P}}}$ represent the mean-reversion rate, the long-run mean and the volatility of the process, respectively.

3 Data and econometric framework

We use daily time series of closing prices of CDSs with maturities 1 year, 3 years and 5 years, for the firms in the ITraxx Europe Series 26, downloaded from Capital IQ, and for these firms the related EDFs at 1 year, 3 years and 5 years, with source Moody's KMV. The sample period goes from 1 January 2007 to 6 February 2017.

From the 125 firms in the iTraxx index, in order to have the entire time series dataset for both CDS prices and EDFs, we end up with 102 firms. Among these, we have 25 financial firms and 77 non financial firms. In terms of credit rating, all of them are investment grade, but 63 have a Moody's rating between Baa3 and Baa1 (and we refer to them as being in the B investment grade class), and 39 are classified by Moody's between A3 and Aaa rating class (and we refer to them as being in the A investment grade class). Given the high variation of CDS prices not only across firms but also in time, instead of reporting static descriptive statistics, we show in Figure 1 the time series of percentiles of 5-year CDS prices, as well as the average prices in the two sectors,

²⁰Empirical studies corroborate the accuracy of EDFs for predicting default (see Kealhofer (2003), Bharath-Shumway (2008), Korablev-Dwyer (2007)) and using EDFs to determine the default premium has been also referred to in Berndt et al. (2005) for US corporate CDS spreads, Vassalou and Xing (2004) for stock prices and Pan and Singleton (2006) for CDS spreads of Japanese banks.

finacial and non financial, and in the two A and B investment grade rating classes. The same information about 5-year EDFs is reported in Figure 2.



Figure 1 – **Time series statistics of 5-year observed CDS prices.** The figure shows in the three panels the observed 5-year CDS prices displayed as percentiles, averages of financial and non financial firms, and averages of A and B investment grade class. The sample period covers from 1 January 2007 to 6 February 2017. The red vertical bar coincides with the date of the first announcement of the ECB related to the CSPP, on 10 March 2016.



Expected Default Frequencies

Figure 2 – Time series statistics of Moody's 5-year expected default frequencies. The figure shows in the three panels the Moody's KMV 5-year expected default frequencies displayed as percentiles, averages of financial and non financial firms, and averages of A and B investment grade class. The sample period covers from 1 January 2007 to 6 February 2017. The red vertical bar coincides with the date of the first announcement of the ECB related to the CSPP, on 10 March 2016.

As in Driessen (2005) and Diaz et al. (2013) the estimation strategy follows a twostep procedure:

- 1. We obtain the maximum likelihood (ML) estimates of the risk-neutral mean of default arrival rates $\lambda_t^{\mathbb{Q}}$ from the CDS spreads. Assuming that the default event is diversifiable, we obtain a estimate of the distress risk premium.
- 2. We estimate the ML actual default intensity $\lambda_t^{\mathbb{P}}$ using the information on the actual probabilities of default embedded in the Moody's KMV EDFs.²¹ We thus compute the jump-at-default risk premium in terms of the ratio $\frac{\lambda_t^{\mathbb{Q}}}{\lambda_t^{\mathbb{P}}}$, as shown by Driessen (2005) and Pan and Singleton (2006).

Our maximum likelihood estimation strategy is based on Chen and Scott (1993),²² and has also been employed in a similar context by Berndt et al. (2005), Pan and Singleton (2008) and Longstaff et al. (2011).

²¹Which are based on the Merton model for picing corporate debt (see Merton (1974)).

 $^{^{22}}$ See also Hamilton and Wu (2012) for a description of the method.

Maximum likelihood estimates of risk-neutral default intensity 3.1

Let us first consider the estimates of $\lambda_t^{\mathbb{Q}}$ from the CDS spreads.

For each time t, we have one unobservable factor, λ_t , and three corresponding CDS prices, with different maturities. Consequently, we must include additional random variables in order to perform a change of variables from the unobservable state variables to the CDS prices. To this aim, we assume CDSs with maturity 3 years (CDS^{3Y}) as being perfectly priced,²³ and we add standard normal measurement errors u^{1Y} and u^{5Y} for CDSs with maturities 1 and 5 years (CDS^{1Y}) and CDS^{5Y} , respectively.

$$CDS_t^{sY} = f^{CDS^{sY}} + u_t^{sY} \quad s = 1, 3, 5$$
 (14)

where

$$\begin{split} u_t^{3Y} &= 0\\ \phi(u_t^{1Y}, u_t^{5Y}) &= \frac{1}{\sqrt{(2\pi)^2 |\Omega|}} e^{-\frac{1}{2} \mathbf{u_t} \prime \Omega^{-1} \mathbf{u_t}} \end{split}$$

and Ω is the diagonal covariance matrix for the measurement errors, with the variances of the measurement errors σ_{1y}^2 and σ_{5Y}^2 on the diagonal.

To build the maximum likelihood estimator for the parameters of our model, we develop a likelihood function for the observed CDS prices as functions of the unobserved deafult intensities. It is important to remark that while for CDS prices we consider the dynamics under the risk neutral measure \mathbb{Q} , when we deal with the time series of default intensities, we must use the dynamics under the objective measure \mathbb{P} . The vector of parameters that we are going to estimate is thus $\Theta = (K^{\mathbb{Q}}, \theta^{\mathbb{Q}}, K^{\mathbb{P}}, \theta^{\mathbb{P}}, \sigma_{\lambda}, \sigma_{1Y}, \sigma_{5Y}).$

We first obtain a possible path of $\lambda_t^{\mathbb{Q}}$ by numerically inverting equation(11).

Given the Ornstein-Uhlenbeck dynamics for $\ln \lambda_t$,²⁴ the loglikelihood for the sample of observations $\log L(\lambda_t, \ldots, \lambda_T)$ is that one of a Gaussian AR(1) process (see Hamilton (1994), page 119).²⁵ Given the relationship (14) and the hypothesis of standard normal

$$\lim_{\Delta t \to 0} \ln \lambda_t - \ln \lambda_{t-1} = \theta(1 - e^{-\kappa}) + (e^{-\kappa} - 1) \ln \lambda_{t-1} + \epsilon_t,$$

 $^{^{23}}$ The same hypothesis is in Longstaff et al. (2011) and Diaz et al. (2013), but is not related to different liquidity consitions of CDS contracts with different maturities.

 $^{^{24}}$ We can consider this dynamics as the continuous time version of the corresponding AR(1) process, or

with $\epsilon_t \sim \mathcal{N}(0, \sigma_\epsilon)$ and $\sigma_\epsilon^2 = \frac{\sigma^2}{2K}(1 - e^{-2K})$. ²⁵A non-central χ^2 distribution, defined in terms of the modified Bessel function of first kind, is instead involved in case of a CIR dynamics for λ_t (see Chen and Scott (1993)).

measurement errors in our model, the log-likelihood function for the entire framework results in the following form:

$$\log L(\lambda_t, \dots, \lambda_T) - \sum_{t=1}^T \left(\log |\mathbf{B}_t| + \log(2\pi) + \frac{1}{2} \log \sigma_{1y}^2 + \frac{1}{2} \log \sigma_{5y}^2 + \frac{(u_t^{1y})^2}{2\sigma^2} + \frac{(u_t^{5y})^2}{2\sigma^2} \right)$$

where the elements of the matrix **B** are such that the Jacobian of the transformation from $\lambda_t^{\mathbb{Q}}$ to CDS_t is $|\mathbf{B}_t|^{-1}$, and σ_{1y} and σ_{5y} are the standard deviations of the measurement errors for the CDSs with maturity 1- and 5-years, respectively.

3.2 Maximum likelihood estimates of actual default intensity

For the estimates of the objective default intensity $\lambda_t^{\mathbb{P}}$ our dataset consists of 1-, 3and 5-year EDFs from Moody's KMV. We employ the same methodology as in Section 3.1, of which in the following we briefly recall the main steps. We assume that 3-year EDFs are observed without error. To estimate the parameters of $\lambda_t^{\mathbb{P}}$ we maximize the likelihood of the joint density of process (13) and the mispricing errors, related to 1- and 5-year EDFs. To this aim we first obtain a possible path of $\lambda_t^{\mathbb{P}}$ by inverting equation(12). Mispricing errors of 1- and 5-year EDFs are normally distributed with zero mean and standard deviations σ_{1y} and σ_{5y} , respectively. To compute the expectation in equation (12) for the intensity process (13) we employ as before the finite difference method of Crank-Nicholson.

4 Estimates of the two risk premia components

In this section we describe the estimation results related to the two sources of excess return.

4.1 Estimation results of the distress risk premium

Table 1 provides summary statistics of maximum likelihood estimates of the parameters of the risk-neutral process $\lambda_t^{\mathbb{Q}}$.

	$K^{\mathbb{Q}}$	$ heta^{\mathbb{Q}}$	σ_{λ}	$K^{\mathbb{P}}$	$ heta^{\mathbb{P}}$	σ_{1y}	σ_{5y}
Mean	0.3288	-4.5333	1.1908	0.4314	-6.6636	0.0016	0.0013
Std	0.0339	0.4705	0.0677	0.0047	0.0717	0.0016	0.0007
Median	0.3294	-4.5036	1.2000	0.4304	-6.6799	0.0012	0.0012

Table 1 – Parameters of risk neutral default intensity process under risk neutral and objective probability measures.

Summary statistics of maximum likelihood estimates of risk-neutral $\lambda_t^{\mathbb{Q}}$ process. $K^{\mathbb{Q}}$ and $K^{\mathbb{P}}$ denote the mean-reversion rates of $\lambda_t^{\mathbb{Q}}$ under the risk-neutral and actual measures, respectively. $\theta^{\mathbb{Q}}$ and $\theta^{\mathbb{P}}$ are the long-run mean of $\lambda_t^{\mathbb{Q}}$ under the risk-neutral and actual measures, respectively. σ_{λ} is the instantaneous volatility of the process. σ_{1y} and σ_{5y} represent the volatility of the mispricing errors for 1- and 5-year maturities.

Looking at these results we can see that on average, mean-reversion rates are higher under actual than under risk-neutral measure $(K^{\mathbb{P}} > K^{\mathbb{Q}})$. Conversely, long-term mean parameters are higher under \mathbb{Q} than \mathbb{P} measure $(\theta^{\mathbb{Q}} > \theta^{\mathbb{P}})$. These results are consistent with empirical evidence in the literature (see, for instance, Pan and Singleton (2008) and Diaz et al. (2013)) and imply that $\lambda^{\mathbb{Q}}$ will tend to be larger under \mathbb{Q} than under \mathbb{P} ; in other words the arrival of credit events is more intense in the risk-neutral (higher long-run means) than in the actual environment. Moreover, for a given level of $\lambda^{\mathbb{Q}}$, there is more persistence under \mathbb{Q} than under \mathbb{P} (bad times last longer in the risk-neutral world as the speed of mean reversion is lower under \mathbb{Q}).

Remembering the choice of the market price of risk Λ_t and the related relationship between the parameters under \mathbb{Q} and \mathbb{P} measures, we can compute the average market price of risk parameters:

$$\gamma_0 = \frac{K^{\mathbb{P}}\theta^{\mathbb{P}} - K^{\mathbb{Q}}\theta^{\mathbb{Q}}}{\sigma_\lambda} = -1.1749$$

and

$$\gamma_1 = \frac{K^{\mathbb{Q}} - K^{\mathbb{P}}}{\sigma_\lambda} = -0.0862$$

These parameters determine a negative Λ_t , for each t: on average we have

$$\bar{\Lambda} = \gamma_0 + \gamma_1 \ln \lambda^{\overline{\mathbb{Q}}} = -0.6699.$$

We can say that the negative sign of coefficients γ_0 and γ_1 confirm that the credit environment is worst under risk-neutral than under actual measure. According to Pan and Singleton (2008), it is this pessimism about the credit environment that allows riskneutral pricing to recover market prices in the presence of investors who are averse to default risk. From the estimates of the parameters of the process $\lambda_t^{\mathbb{Q}}$ under both \mathbb{Q} and \mathbb{P} measures, we can compute the distress risk premium as in equation (10).

In Figure 3 are displayed the average 5 year CDS prices estimated under the riskneutral measure \mathbb{Q} and the objective measure \mathbb{P} , and the observed prices. The green line, representing the CDS price under the objective measure, provides the time series of the estimated expected losses.



Figure 3 – Average 5-year CDS prices observed and estimated under the two probability measures.

The figure shows the average 5-year CDS prices, estimated under the \mathbb{Q} and \mathbb{P} measures, and observed. The sample period covers from 1 January 2007 to 6 February 2017. The red vertical bar coincides with the date of the first announcement of the ECB related to the CSPP, on 10 March 2016.

Figure 4 shows the estimated average and the percentiles of the distress risk premia for the firms in the iTraxx index in basis points.



Figure 4 – Distress risk premium: average and percentiles.

The figure shows the average and percentiles of the distress risk premium embedded in 5-year CDS prices. The sample period covers from 1 January 2007 to 6 February 2017. The red vertical bar coincides with the date of the first announcement of the ECB related to the CSPP, on 10 March 2016.

According to the intuition and the empirical evidence in the literature, we can see that the distress risk premium increases markably during the periods of the financial crisis and the sovereign debt crisis. Moreover there is a considerable dispersion between the firms in the index, as we can see when looking at the percentiles.

Figure 5 displays the different behaviour of the distress risk premia for the financial and non financial firms and according to the rating class.



Figure 5 – Distress risk premium for financial and non financial firms and for rating class.

The figure shows the average distress risk premium, embedded in 5-year CDS prices, in the financial and non financial subsets and in two rating classes subsets. The sample period covers from 1 January 2007 to 6 February 2017. The red vertical bar coincides with the date of the first announcement of the ECB related to the CSPP, on 10 March 2016.

We can say that while both financial and non financial firms were deeply affected by the financial crisis, the sovereign debt crisis hit most the financial firms. Finally, as we would expect, the distress risk premium of the firms in the lower rating classes is higher than in the higher rating classes.

4.2 Estimation results of the jump-at-default risk premium

Table 2 displays the average parameters of the actual default instensity process $\lambda_t^{\mathbb{P}}$:

	$\alpha^{\mathbb{P}}$	$\beta^{\mathbb{P}}$	σ_{λ}	σ_{1y}	σ_{5y}
Mean	0.3035	-12.5442	1.1360	0.0024	0.0012
Std.	0.0948	12.6319	0.1721	0.0046	0.0016
Median	0.3079	-9.3997	1.1735	0.0011	0.0006

Table 2 – Parameters of objective default intensity process.

Summary statistics of maximum likelihood estimates of actual $\lambda_t^{\mathbb{P}}$ process. $K^{\mathbb{P}}$ denotes the meanreversion rate of $\lambda_t^{\mathbb{P}}$ under the actual measure. $\theta^{\mathbb{P}}$ is the long-run mean of $\lambda_t^{\mathbb{P}}$ under the actual measure. σ_{λ} is the instantaneous volatility of the process. σ_{1y} and σ_{5y} represent the volatility of the errors in the EDFs for 1- and 5-year maturities.

In Figure 6 are displayed the average 5-year EDFs reported by Moody's KMV and estimated with our model.



Figure 6 – Average 5-year EDFs of source Moody's and estimated.

The figure shows the average 5-year EDFs of source Moody's KMV and estimated. Data are in percentages. The sample period covers from 1 January 2007 to 6 February 2017. The red vertical bar coincides with the date of the first announcement of the ECB related to the CSPP, on 10 March 2016.

Once we have estimated the actual default intensities, we can compute the jumpat-default premium $\lambda_t^{\mathbb{Q}}/\lambda_t^{\mathbb{P}}$ as well as the market price of jump-at-default risk Γ_t . The jump-at-default premium ratio results in approximately 2.9 on (averaged) median for the firms under study. This estimate is consistent with those previously reported in the literature for the US market (see, among others, Driessen (2005)) and for the European market (see for instance Diaz et al. (2013)). Employing the definiton of jump-a-defaultrisk premium in equation (1), and having assumed a recovery rate of 40%, we have

$$JADRP_t = (0.4 - 1) * \lambda_t^{\mathbb{P}} * (1 - \frac{\lambda_t^{\mathbb{Q}}}{\lambda_t^{\mathbb{P}}})$$
(15)

Figures 7 and 8 show the time-series of jump-at-default risk premium in basis points, displayed as average and percentiles, and divided in the two classes, financial and non financial firms and by rating class, respectively. Our (averaged) median jump-at-default estimate is 11 basis points, that is in line with the existing literature (Diaz et al. (2013) estimate a (averaged) median jump-at-default of 13 basis points). This result indicates that the default event itself carries a risk premium and that this premium increases during the stress periods, which agrees the intuition about the jump-at-default risk premium.



Figure 7 – **Jump-at-default risk premium: average and percentiles.** The figure shows the average and the percentiles of the jump-at-default risk premium. Data are in basis points. The sample period covers from 1 January 2007 to 6 February 2017. The red vertical bar coincides with the date of the first announcement of the ECB related to the CSPP, on 10 March 2016.



Figure 8 – Jump-at-default risk premium for financial and non financial firms and for rating class.

The figure shows the average jump-at-default risk premium in the financial and non financial subsets and in the two rating classes subsets. Data are in basis points. The sample period covers from 1 January 2007 to 6 February 2017.

It is evident that the average jump at default risk premium is slightly negative in calm periods. In fact, according to formula (15), jump at default risk premium is negative if $\lambda^{\mathbb{P}} > \lambda^{\mathbb{Q}}$, and this happens, on average, during calm periods, as displayed in Figure 9.



Default instensity under \mathbb{P} and \mathbb{Q} measures (per cent)

Figure 9 – Average physical and risk-neutral default intensity. The figure shows the average of the estimated default intensity under \mathbb{P} and \mathbb{Q} measures, in percentages. The sample period covers from 1 January 2007 to 6 February 2017. The red vertical bar coincides with the date of the first announcement of the ECB related to the CSPP, on 10 March 2016.

The jump-at-default-risk-premium is not statistically different from zero during the calm periods. Negative values might arise from the problem of realibility of $\lambda^{\mathbb{P}}$ estimates, a problem that persists in the existing literature: in fact, the estimation of the actual default intensity is a really difficult task, and we do not count on optimal estimates of it. However, we employ EDFs because they have a higher predictive power than credit ratings, depending on stock prices and being time-variant. Several empirical studies corroborate the accuracy of EDFs for predicting default.²⁶ Additionally, using EDFs to determine the default premium has also been referred to in Berndt et al. (2005) for US corporate CDS spreads, in Vassalou and Xing (2004) for stock prices and in Pan and Singleton (2006) for the CDS spreads of Japanese banks.

The problem of negative jump-at-default risk premium disappears when looking at the median, as we can see in Figure 7.

²⁶See, for instance, Kealhofer (2003), Bharath and Shumway (2008) and Korablev and Dwyer (2007).

5 Analysis of the two risk premium components and event study exercise on the risk premia and the expected losses

Putting together the two estimated risk premia, the distress risk premium and the jumpat-default risk premium, we obtain the excess return of the corporate bond:

$$e_t = DRP_t + JADRP_t$$

Figure 10 displays the time series of the estimated average excess return.





The figure shows the average excess return, obtained as the sum of the distress risk premium and the jump-at-default risk premium. The sample period covers from 1 January 2007 to 6 February 2017. The red vertical bar coincides with the date of the first announcement of the ECB related to the CSPP, on 10 March 2016.

According to our estimates, the (averaged) median distress risk premium in the entire sample period is around 60 basis points and accounts for 88% of the excess return, while the remaining (averaged) median jump-at-default risk premium is around 11 basis points and accounts for the residual 12%. If we look at the most recent period, the contribute of the distress risk premium is even higher: in the data sample starting on January 2016 the (averaged) median distress risk premium is around 53 basis points and accounts for 93%, while the (averaged) median jump-at-default risk premium is around 3 basis points and accounts for 7%. An interesting comparison is between the behaviour of the two risk premium components in the financial and non financial sectors. Looking at the same recent period, we can say the following:

- The (averaged) median distress risk premium is higher for the financial firms (around 69 basis points, versus 49 basis points of the non financial counterparties);
- On the contrary, the (averaged) median jump-at-default risk premium is higher for the non financial firms (being around 4 basis points, versus a negative value of the financial counterparties);
- There is a higher heterogeneity between financial financial firms with respect to the non financial counterparties, as both the dispersion in the distress risk premium and in the jump-at-default risk premium are higher for the financial firms: in particular the average standard deviation of the distress risk premium is around 30 basis points for the financial firms, versus 22 for the non financial; the average standard deviation of the jump-at-default risk premium is around 73 basis points, versus 19 for the non financial firms.

We conduct an event study exercise around key ECB announcements related to the Corporate Sector Purchase Programme (CSPP) during the press releases to investigate the effects of these announcements, and thus of the launch of this programme, in the decrease of the two risk premium components. We consider the following dates:²⁷

- 10 March 2016: ECB adds corporate sector purchase programme (CSPP) to the asset purchase programme (APP);²⁸
- 21 April 2016: Press Release: ECB announces details of the corporate sector purchase programme (CSPP), mainly on the eligibility of bonds;²⁹
- 2 June 2016: ECB announces remaining details of the corporate sector purchase programme (CSPP) on the issuer eligibility.³⁰

²⁷Of course we expect that the main impact on the components of CDS prices arises at the date of the first announcement of the CSPP (as in fact is confirmed by the results), but we also consider the dates of the other two press releases in which the ECB provides details of the programme.

²⁸See http://www.ecb.europa.eu/press/pr/date/2016/html/pr160310_2.en.html.

²⁹See http://www.ecb.europa.eu/press/pr/date/2016/html/pr160421_1.en.html.

 $^{^{30}} See \ \texttt{https://www.ecb.europa.eu/press/pr/date/2016/\texttt{html/pr160602_1.en.html}.$

We use a time window of ten days, five before and five after the announcements. Unfortunately, we cannot use a difference in difference methodology³¹ to adjust for changes in financial sector risk because also corporate spreads of financial institutions benefited from ECB decisions in those dates. In terms of rating, all the bonds of the firms in the iTraxx are eligible for the purchases in the CSPP, having an investment grade rating.

Figure 11 shows the median of the corporate bond excess return around announcement dates, while Figure 12 shows separately the two components of the excess return - the median of the distress risk premium and the median of the jump-at-default risk premium - around the same dates. Looking at these figures it is evident that most of the decrease in both risk premia was observed on 10 March 2016. In fact, also in the ECB Economic Bulletin (Issue 5, 2016) is pointed out that "the announcement of the CSPP on 10 March was followed by a significant contraction in the spread between yields on bonds issued by non-financial corporations (NFCs) and a risk-free rate". A further spread contraction can be observed on 21 April (when was confirmed the eligibility of insurance corporations), while in June, spreads developments where also related to the uncertainty generated by the UK referendum. Table 3 displays the average variations of the excess return and its components between the day after the ECB announcement on 10 March and the day before, while in Table 4 we can see the historical daily variations. Looking at the different magnitudes of the variations reported in the two tables, we can say that the reductions observed on 10 March where significant.

$\Delta E \bar{x} Ret$	$\Delta D \overline{R} P$	$\Delta J A \overline{D} R P$
-16.75	-10.87	-5.88

Table 3 – Average variations around ECB announcements

This table shows the average variations between the day after the ECB announcement on 10 March 2016 and the day before, of the excess return, the distress risk premium and the jumpat-default risk premium. Basis points.

$h\Delta \bar{ExRet}$	$h\Delta \bar{D}RP$	$h\Delta J\bar{A}DRP$
0.010	0.009	0.001

Table 4 – Historical daily variations

This table shows the historical daily variations of the excess return, the distress risk premium and the jump-at-default risk premium. Basis points.

³¹As often employed in the literature for event study exercises, see for instance Kelly at al. (2012).

Table 5 reports the percentage variations of the excess return and its components between the day after the ECB announcement on 10 March and the day before.

$(\Delta ExRet)/ExRet_{t-1}$	$(\Delta DRP)/DRP_{t-1}$	$(\Delta JADRP)/JADRP_{t-1}$
-21.41	-15.84	-34.47

Table 5 – **Average relative variations around ECB announcements** This table shows the average relative variations between the day after the ECB announcement on 10 March 2016 and the day before, of the excess return, the distress risk premium and the jump-at-default risk premium. Percentages.

The decreases in the distress risk premium and the jump-at-default risk premium were on average of 15.8 and 34.5 per cent, respectively, between the day following this announcement and the day before. We can thus say that jump-at-default component was more affected, in relative terms, by the announce of the CSPP.





Figure 11 – Event study: median of the excess return.

The figure shows the median of the corporate bond excess return around announcement dates. The black line shows the average risk premium response over all announcements, and the coloured lines show the responses to each individual announcement.



Risk premia around ECB announcements (basis points)

Figure 12 – **Event study: median of the distress and jump-at-default risk premia.** The figure shows the median of the distress risk premium and the jump-at-default risk premium around announcement dates. In each figure, the black line shows the average risk premium response over all announcements, and the coloured lines show the responses to each individual announcement.

With the same event study exercise we can also analyze the different responses of the financial and non financial sectors to the CSPP announcements.

Figure 14 shows the median of the corporate bond excess return of the financial and non financial sectors, around announcement dates, while Figure 15 shows separately the two components of the excess return - the median of the distress risk premium and the median of the jump-at-default risk premium in the two sectors - around the same dates.

Table 6 displays the average variations of the excess return and its components, for the financial and non financial sectors, between the day after the ECB announcement on 10 March and the day before; in Table 7 we can see the related percentage variations.

$\Delta ExRetf$	$\Delta D \overline{R} P f$	$\Delta JA\bar{D}RPf$	$\Delta Ex \overline{R}etnf$	$\Delta D\bar{R}Pnf$	$\Delta JA\bar{D}RPnf$
-11.75	-13.14	-1.87	-16.68	-7.12	-4.71

Table 6 – Average variations around ECB announcements in the financial and non financial sectors

This table shows the average absolute variations between the day after the ECB announcement on 10 March 2016 and the day before, of the excess return, the distress risk premium and the jump-at-default risk premium in the financial and non financial sectors. Basis points.

$\frac{\Delta ExRetf}{ExRetf_{t-1}}$	$\frac{\Delta DRPf}{DRPf_{t-1}}$	$\frac{\Delta J A \overline{D} R P f}{J A D R P f_{t-1}}$	$\frac{\Delta ExRetnf}{ExRetnf_{t-1}}$	$\frac{\Delta D\bar{R}Pnf}{DRPnf_{t-1}}$	$\frac{\Delta J A \overline{D} R P n f}{J A D R P n f_{t-1}}$
-26.43	-15.67	-10.06	-25.20	-12.24	-50.92

Table 7 – Average relative variations around ECB announcements in the financial and non financial sectors

This table shows the average relative variations between the day after the ECB announcement on 10 March 2016 and the day before, of the excess return, the distress risk premium and the jump-at-default risk premium in the financial and non financial sectors. Percentages.

What is evident is that the spillovers of the launch of the CSPP to the financial sector have been rather important: in relative terms the drop of the excess return of the financial sector has even been slightly higher than in the non financial sector (-26% vs -25%), as well as the decrease of the distress risk premium component; the percentage variation of the jump-at-default risk premium component has instead been more pronounced in the non financial sector.

To verify that the changes in the estimated risk premia components after the announcements were substantial, Table 8 displays the historical daily variations of the excess return and its components for the financial and non financial sectors, and we can confirm that they are significantly smaller.

$h\Delta E\bar{x}Retf$	$h\Delta D\bar{R}Pf$	$h\Delta JA\overline{D}RPf$	$h\Delta ExRetnf$	$h\Delta D\bar{R}Pnf$	$h\Delta JA\bar{D}RPnf$
0.017	0.019	-0.002	0.007	0.005	0.002

Table 8 – Historical daily variations in the financial and non financial sectors This table shows the average historical daily variations of the excess return, the distress risk premium and the jump-at-default risk premium in the financial and non financial sectors. Basis points.

In addition to the risk premia, we know that a CDS price embeds the compensation for the expected losses. If we also add the estimates of the expected losses, we obtain the decomposition of the CDS price into its three components. Figure 13 displays at selected dates the average estimates of the expected losses, the distress risk premium and the jump-at-default risk premium.



Decomposition of the average CDS price (basis points)

Figure 13 – Components of average CDS price.

The figure shows the three components of the average CDS price (the average expected losses, the average distress risk premium and the average jump-at-default risk premium) at selected dates: before the financial crisis (January 2007), at the peaks of the financial crisis (January 2009) and of the sovereign debt crisis (January 2012); before the announcement of the programme (March 2016); when the purchases begin (June 2016); at the most recent date (February 2017).

We can also estimate the effect of the CSPP on the expected losses component of the CDS prices, by looking at the variation of the CDS price under the \mathbb{P} measure: on average, between the day before the CSPP announcement and the day after, the expected losses decreased by 3.7 basis points; the reduction for the financial firms was more pronounced than for the non financial counterparties (5.2 versus 3.2 basis points). These variations are quite important, given that the historical daily variations of the expected losses components are 0.003 basis points on average for the entire sample and 0.006 and 0.002 basis points for the financial and non financial firms, respectively.

Excess return around ECB announcements in the financial and non financial sectors (basis points)

Figure 14 – Event study: median of the excess return in the financial and non financial sectors.

The figures show the median of the corporate bond excess return around announcement dates in the financial and non financial sectors. In each figure the black line shows the average risk premium response over all announcements, and the coloured lines show the responses to each individual announcement.

Risk premia around ECB announcements in the financial and non-financial sectors



Figure 15 – Event study: median of the distress and jump-at-default risk premia in the financial and non-financial sectors.

The figures show the median of the distress risk premium and the jump-at-default risk premium around announcement dates. In each figure, the black line shows the average risk premium response over all announcements, and the coloured lines show the responses to each individual announcement.

6 Conclusion

Among the unconventional monetary policy measures recently adopted by the ECB, the Corporate Sector Purchase Programme (CSPP) has especially exerted a downward pressure on the corporate bond spreads. We contribute to the recent literature on the assessment of the effectiveness of unconventional monetary policy measures, with a focus on the corporate bond market. Corporate bonds are carefully monitored by central banks because they constitute an important channel of transmission of monetary policy decisions to the real economy.

The purpose of this paper is to investigate if the reduction observed in corporate bond spreads after the launch of the programme can be attributed more to a decrease of the expected losses or to a reduction of the risk premia, and in particular the compensation required by investors for changes in the the credit environment associated with business and macro conditions and the compensation for the risk associated with a jump in price of the bond in case of default.

To this aim, following Diaz et al. (2013), we use the information embedded in the Credit Default Swaps of the firms in the iTraxx Europe index, and the corresponding Expected Default Frequencies, to decompose the excess return of the corporate bond of each firm into two sources of risk premia: the distress risk premium and the jumpat-default risk premium. Our estimates are in line with those previously reported in the literature. In relative terms, we find that the distress risk premium accounts for approximately 88% of the total excess return on (averaged) median, historically. On the other end, the jump-at-default risk premium accounts for the remaining 12% on (averaged) median. If we restrict our time horizon to the most recent period (since January 2016), the relative contribute of the distress risk premium has even increased, reaching the 93% of the excess return. The estimated risk premia show significant and interesting variations not only in time, with marked increases in both risk premium components during the two main crisis periods of the last ten years (the financial crisis and the sovereign debt crisis), but also across type of firms (although the bonds are all investment grade rated) and sectors. In particular, in the recent period the dispersion of both risk premium components is higher for financial firms; moreover, this sector exhibits a higher distress risk premium, while the jump-at-default risk premium is more pronounced for non-financial firms.

Our analysis also allows us to estimate the expected losses component of the CDS price, provided by the CDS price obtained under the objective measure (that does not reflect the investors' risk preferences).

To assess on which source of risk premium the CSPP has been more effective, we

perform an event study exercise around the dates of the announcements of the ECB related to the launch of this programme and we can see, as we would expect, that the date correspondent to the main decrease of the CDS price and its components, is the date of the first announcement of the programme, on 10 March 2016. Looking at the variations between the day before and the day after the first announcement of the programme, we find that on average the decrease of the risk premia was about 16.7 basis points $(-10.9 \text{ and } -5.8 \text{ basis points for the distress risk premium and the jump-at$ default risk premium, respectively), while the decrease of the expected losses was about 3.7 basis ponts, suggesting a more important effect of the CSPP on the risk premia component, and in particular on the distress risk premium component; however, also the reduction of the expected losses component is not negligible, proving that the CSPP has also contributed to diminuish the default probability of the issuers, by improving the financial conditions and the expectations on the economic outlook. From the same exercise we also identify important spillovers of the launch of the CSPP to the financial sector, in particular in terms of a decrease of the distress risk premium and of the expected losses.

Subject of future research will be an analysis of the systemic and idiosycratic risk in corporate bond market, based on our firm-by-firm analysis and a comparison of the results obtained calibrating the model to the overall iTraxx index with those of a theoretical iTraxx index obtained from the aggregation of individual CDSs.

Appendix

A.1 Decomposition of the excess return into two components

In this appendix we provide mathematical evidence to explain the different sources of the excess return of a defaultable security. One first approach consists of looking at the equation for the price dynamics under actual and risk neutral measures.³² The price of a default-risky bond can be considered to depend on a set of state variables X_t the follow a diffusion process

$$dX_t = \mu_X^{\mathbb{P}}(X_t, t)dt + \sigma_X(X_t, t)dW_t^{\mathbb{P}}$$
(A.1.16)

where $\mu_X^{\mathbb{P}}$ and σ_X are the drift and instantaneous volatility, respectively, under the actual measure \mathbb{P} . According to Girsanov's theorem the same equation can be represented under the risk-neutral measure \mathbb{Q} :

$$dX_t = \mu_X^{\mathbb{Q}}(X_t, t)dt + \sigma_X(X_t, t)dW_t^{\mathbb{Q}}$$
(A.1.17)

where

$$W_t^{\mathbb{Q}} = W_t^{\mathbb{P}} + \int_0^t \Lambda_t ds,$$

and

$$\mu_X^{\mathbb{Q}} = \mu_X^{\mathbb{P}}(X_t, t) - \sigma_X(X_t, t)\Lambda_t$$

are the Brownian motion and the drift under the risk-neutral measure \mathbb{Q} , with Λ_t being the price of risk.

The dynamics of the price $P(t, X_t)$ can be represented by applying Ito's lemma³³ under \mathbb{Q} measure:

$$dP(X_t, t) = \frac{\partial P(X_t, t)}{\partial t} dt + \frac{\partial P(X_t, t)}{\partial X_t} r_t dt + \frac{\partial P(X_t, t)}{\partial X_t} \sigma_X(X_t, t) dW_t^{\mathbb{Q}}$$
(A.1.18)

$$+\frac{1}{2}\frac{\partial^2 P(X_t,t)}{\partial X_t^2}\sigma_X^2(X_t,t)dt + (R_t - P(X_t,t))\lambda_t^{\mathbb{Q}}dt$$
(A.1.19)

and under $\mathbb P$ measure:

$$dP(X_t, t) = \frac{\partial P(X_t, t)}{\partial t} dt + \frac{\partial P(X_t, t)}{\partial X_t} (r_t + \sigma_X(X_t, t)\Lambda_t) dt + \frac{\partial P(X_t, t)}{\partial X_t} \sigma_X(X_t, t) dW_t^{\mathbb{P}}$$
(A.1.20)

$$+\frac{1}{2}\frac{\partial^2 P(X_t,t)}{\partial X_t^2}\sigma_X^2(X_t,t)dt + (R_t - P(X_t,t))(\lambda_t^{\mathbb{Q}} - \psi(t)\lambda_t^{\mathbb{P}})dt$$
(A.1.21)

³²Technical details can be also found in Cont-Tankov (2004).

³³Terms come from Ito's lemma for a process with drift, diffusion and Poisson jumps (due to default) and Girsanov's theorem; see A.2 for a brief overview and Cont-Tankov (2004) for details.

with $\psi(t) = \frac{\lambda^{\mathbb{Q}}}{\lambda^{\mathbb{P}}} - 1$. The expected excess return of the risky-bond can be defined as the difference of expectations under \mathbb{Q} and \mathbb{P} measures:

$$\mathbb{E}^{\mathbb{P}}\Big[\frac{dp(X_t,t)}{P(X_t,t)}\Big] - \mathbb{E}^{\mathbb{Q}}\Big[\frac{dp(X_t,t)}{P(X_t,t)}\Big] = \frac{1}{P_t}\frac{\partial P_t}{\partial X_t}\sigma_X(X_t,t)\Lambda_t + \frac{P(X_t,t) - R_t}{P(X_t,t)}\lambda_t^{\mathbb{P}}\psi_t \quad (A.1.22)$$

A.2 Ito's lemma for a process with drift, diffusion and Poisson jumps

In this appendix we just want to give the main intuition/ingredients underlying the Ito's lemma when we have a process that includes jumps.³⁴ Consider a Poisson process with jump intensity λ_1 .³⁵ The probability of one jump in the interval $[t, t + \Delta t]$ is $\lambda_1 \Delta t$. The survival probability $\bar{p}(t)$ (i.e. the probability that no jump has occurred in the interval [0, t]) is

$$\bar{p}(t) = e^{-\int_0^t \lambda_1(s)ds}.$$

Let X(t) be a discontinuous stochastic process and denote $X(t^{-})$ the value of X as we approach t from the left. Denote by $d_j X(t)$ the non-infinitesimal change in X(t) as a result of a jump. Then

$$d_j X(t) = \lim_{\Delta t \to 0} (X(t + \Delta t) - X(t^-)).$$

Let z be the magnitude of the jump and let $\eta(X(t^{-}), z)$ be the distribution of z. The expected magnitude of the jump is

$$\mathbb{E}[d_j(X(t)] = \lambda_1(X(t^-))dt \int_z z\eta(X(t^-), z)dz.$$

Define $dJ_X(t)$, a compensated process and martingale, as

$$dJ_X(t) = d_j X(t) - \mathbb{E}[d_j X(t)].$$

Then

$$d_j X(t) = \mathbb{E}[d_j X(t)] + dJ_X(t) = \lambda_1(X(t^-)) \left(\int_z z\eta(X(t^-), z) dz \right) + dJ_X(t) + dJ_X($$

Consider now a function f(X(t),t) of the jump process dX(t). If X(t) jumps by Δx then f(X(t)) jumps by Δf . Δf is drawn from distribution $\eta_f()$. The jump part of f is

$$f(X(t)) - f(X(t^{-})) = \lambda_1(t)dt \int_{\Delta f} \Delta f \eta_f(\cdot) d\Delta f + dJ_f(t).$$

³⁴Technical details can be found in Cont-Tankov (2004).

 $^{^{35}\}lambda_1$ could be a constant, a deterministic function of time, or a stochastic process.

Let X_t be a diffusion process with jumps, defined as the sum of a drift term, a Brownian stochastic integral and a compound Poisson process:

$$X_{t} = X_{0} + \int_{0}^{t} \mu_{s} ds + \int_{0}^{t} \sigma_{s} dW_{s} + \sum_{i=1}^{N_{t}} \Delta X_{i}.$$

Then, the process $f(t, X_t)$ can be represented in differential notation as:

$$df(t, X_t) = \frac{\partial f}{\partial t}(t, X_t)dt + \mu_t \frac{\partial f}{\partial x}(t, X_t)dt + \frac{\sigma_t^2}{2} \frac{\partial^2 f}{\partial x^2}(t, X_t)dt$$
(A.2.23)

$$+\frac{\partial f}{\partial x}(t,X_t)\sigma_t dW_t + [f(X_{t-} + \Delta X_t) - f(X_{t-})].$$
(A.2.24)

The term linked to the Girsanov transform is related to the derivative term (in addition to the usual derivative in case of a diffusion process) that comes from the change of the intensity of jumps: to change from a Pisson variable N of parameter λ_1 to a variable of parameter λ_2 , the derivative is

$$dP_2/dP_1(N) = (\lambda_2/\lambda_1)^N \exp{-(\lambda_2 - \lambda_1)}.$$

A.3 Ornstein Uhlenbeck dynamics for the risk neutral default intensity under the objective measure

Consider the Ornstein Uhlenbeck dynamics for the logarithm of the default intensity λ_t under the risk neutral measure \mathbb{Q} :

$$d\ln\lambda_t^{\mathbb{Q}} = K^{\mathbb{Q}}(\theta^{\mathbb{Q}} - \ln\lambda_t^{\mathbb{Q}})dt + \sigma_\lambda dW_t^{\mathbb{Q}}$$
(A.3.25)

Assume a market price of risk of the form

$$\Lambda_t = \gamma_0 + \gamma_1 \ln \lambda_t$$

We can write the relationship between the Wiener process under the risk neutral measure $W_t^{\mathbb{Q}}$ and the Wiener process under the objective measure $W_t^{\mathbb{P}}$ as

$$W_t^{\mathbb{Q}} = W_t^{\mathbb{P}} + \int_0^t \gamma_0 + \gamma_1 \ln \lambda_s ds$$

and, in differential terms,

$$dW_t^{\mathbb{Q}} = dW_t^{\mathbb{P}} + (\gamma_0 + \gamma_1 \ln \lambda_t) dt.$$

We can thus express (A.3.25) in terms of $W_t^{\mathbb{Q}}$ as

$$d\lambda_t = (K^{\mathbb{Q}}\theta^{\mathbb{Q}} + \sigma\gamma_0 - (K^{\mathbb{Q}} - \sigma\gamma_1)\ln\lambda_t)dt + \sigma_\lambda dW_t^{\mathbb{P}}$$
(A.3.26)
$$= K^{\mathbb{P}}(\theta^{\mathbb{P}} - \ln\lambda_t)dt + \sigma_\lambda dW_t^{\mathbb{P}}$$

where

$$K^{\mathbb{P}}\theta^{\mathbb{P}} = K^{\mathbb{Q}}\theta^{\mathbb{Q}} + \sigma_{\lambda}\gamma_{0};$$
$$K^{\mathbb{P}} = K^{\mathbb{Q}} - \sigma_{\lambda}\gamma_{1}.$$

A.4 CIR dynamics for the risk neutral default intensity under the objective measure

Consider the CIR dynamics for the default intensity λ_t under the risk neutral measure \mathbb{Q} :

$$d\lambda_t = K^{\mathbb{Q}}(\theta^{\mathbb{Q}} - \lambda_t)dt + \sigma_\lambda \sqrt{\lambda_t} dW_t^{\mathbb{Q}}$$
(A.4.27)

 $K^{\mathbb{Q}}$ is the speed of adjustment, and must be strictly positive; to assure a $\lambda_t^{\mathbb{Q}}$ strictly positive we must also impose the Feller condition: $2K^{\mathbb{Q}}\theta^{\mathbb{Q}} > \sigma_{\lambda}^2$.

We assume a market price of risk as in Cheridito et al. (2007)

$$\Lambda_t = \frac{\gamma_0}{\sqrt{\lambda_t}} + \gamma_1 \sqrt{\lambda_t}$$

We can write the relationship between the Wiener process under the risk neutral measure $W_t^{\mathbb{Q}}$ and the Wiener process under the objective measure $W_t^{\mathbb{P}}$ as

$$W_t^{\mathbb{Q}} = W_t^{\mathbb{P}} + \int_0^t \left(\frac{\gamma_0}{\sqrt{\lambda_s}} + \gamma_1 \sqrt{\lambda_s}\right) ds$$

and, in differential terms,

$$dW_t^{\mathbb{Q}} = dW_t^{\mathbb{P}} + \frac{\gamma_0}{\sqrt{\lambda_t}}dt + \gamma_1\sqrt{\lambda_t}dt.$$

We can thus express (A.4.27) in terms of $W_t^{\mathbb{Q}}$ as

$$d\lambda_t = K^{\mathbb{Q}}\theta^{\mathbb{Q}} + \sigma\gamma_0 - (K^{\mathbb{Q}} - \sigma\gamma_1)\lambda_t dt + \sigma_\lambda \sqrt{\lambda_t} dW_t^{\mathbb{P}}$$
(A.4.28)
$$= K^{\mathbb{P}}(\theta^{\mathbb{P}} - \lambda_t)dt + \sigma_\lambda \sqrt{\lambda_t} dW_t^{\mathbb{P}}$$

where

$$K^{\mathbb{P}}\theta^{\mathbb{P}} = K^{\mathbb{Q}}\theta^{\mathbb{Q}} + \sigma_{\lambda}\gamma_{0};$$
$$K^{\mathbb{P}} = K^{\mathbb{Q}} - \sigma_{\lambda}\gamma_{1}.$$

In order to satisfy the Novikov condition we must impose $\gamma_0 \leq K_0^{\mathbb{P}} \theta^{\mathbb{P}} - \frac{1}{2}$.

A.5 Explicit solution for the CDS price in the CIR model

Under the risk neutral measure \mathbb{Q} consider the equation for the CDS price

$$CDS_t^{\mathbb{Q}}(M) = \frac{(1-R^{\mathbb{Q}})\int_t^{t+M} D(t,u)\mathbb{E}_t^{\mathbb{Q}}\left[\lambda_u^{\mathbb{Q}}e^{-\int_t^u \lambda_s^{\mathbb{Q}}ds}\right]du}{\int_t^{t+M} D(t,u)\mathbb{E}_t^{\mathbb{Q}}\left[e^{-\int_t^u \lambda_s^{\mathbb{Q}}ds}\right]du}$$
(A.5.29)

and the CIR dynamics for $\lambda_t^{\mathbb{Q}}$ as in (9)

We can explicitly solve the two expectations in $(A.5.29)^{36}$ as

$$\mathbb{E}_{t}^{\mathbb{Q}} \left[\lambda_{u}^{\mathbb{Q}} e^{-\int_{t}^{u} \lambda_{s}^{\mathbb{Q}} ds} \right] = \left(K^{\mathbb{Q}} \theta^{\mathbb{Q}} A(u-t) + \frac{\partial A(u-t)}{\partial u} \lambda_{t}^{\mathbb{Q}} \right) \cdot C(u-t) e^{-A(u-t)\lambda_{t}^{\mathbb{Q}}}$$
$$\mathbb{E}_{t}^{\mathbb{Q}} \left[e^{-\int_{t}^{u} \lambda_{s}^{\mathbb{Q}} ds} \right] = C(u-t) e^{-A(u-t)\lambda_{t}^{\mathbb{Q}}}$$

where

$$A(u-t) = \frac{2(e^{\gamma(u-t)}-1)}{(\gamma+K^{\mathbb{Q}})(e^{\gamma(u-t)}-1)+2\gamma}$$
$$\frac{\partial A(u-t)}{\partial u} := B(u-t) = \frac{4\gamma^2 e^{\gamma(u-t)}}{((\gamma+K^{\mathbb{Q}})(e^{\gamma(u-t)}-1)+2\gamma)^2}$$
$$C(u-t) = \left[\frac{2\gamma e^{\frac{1}{2}(\gamma+K^{\mathbb{Q}})(u-t)}}{(\gamma+K^{\mathbb{Q}})(e^{\gamma(u-t)}-1)+2\gamma}\right]^{\frac{2K^{\mathbb{Q}}\theta^{\mathbb{Q}}}{\sigma_{\lambda}^2}}$$

and

$$\gamma = \sqrt{K^{\mathbb{Q}^2} + 2\sigma_\lambda^2}.$$

A.6 PDE associated to the expectation and Crank Nicholson scheme

Consider a stochastic process

$$dy_t = p(y,t)dt + \sigma(y,t)dW_t$$

with W_t standard Brownian motion, and the partial differential equation

$$S_t + S_y p(y,t) + \frac{1}{2} S_{yy} \sigma(y,t)^2 = c(y,t)S$$
(A.6.30)

with boundary condition g(y,T) at T, where p(y,t) and c(y,t) are two functions of y and t. According to the Feynman-Kac theorem, the solution of equation A.6.30 is given by

$$S(y,T) = \mathbb{E}^{\mathbb{Q}}\left[e^{-\int_t^T c(y_u,u)du}g(y_T,T)\right].$$
(A.6.31)

 $^{^{36}}$ Technical details can be found in Duffie and Singleton (2003) and Schonbucher (2003).

In our paper we want to compute $\mathbb{E}^{\mathbb{Q}}\left[e^{-\int_{t}^{u}\lambda_{s}ds}\right]$ so we set

$$y_t := x_t = \ln \lambda_t$$
$$p(y,t) := K\theta - Kx_t$$
$$\sigma(y,t) := \sigma$$
$$c(y_u, u) := \lambda_s = e^{x_s}$$
$$g(y_T, T) := 1.$$

Our expectation is then solution of the following PDE:

$$S_t + S_x(K\theta - Kx_t) + \frac{1}{2}S_{xx}\sigma^2 = e^{x_t}S$$
 (A.6.32)

with boundary condition S(x, u) = 1.

At this point we can employ the Crank-Nicholson scheme to solve it. This is a finite difference method that consists of replacing terms in equation A.6.30 by approximation in discrete time and space. Consider: N time steps with $t_i - t_{i-1} = \Delta t$ constant; L space steps with $x_j - x_{j-1} = \Delta x$ constant. We can approximate the derivatives in time and space as

$$\begin{split} \frac{\partial S}{\partial t} &\approx \frac{S_{j,i+1} - S_{j,i}}{\Delta t} \\ \frac{\partial S}{\partial x} &\approx \frac{1}{2} \frac{S_{j+1,i} - S_{j-1,i}}{2\Delta x} + \frac{1}{2} \frac{S_{j+1,i+1} - S_{j-1,i+1}}{2\Delta x} \\ \frac{\partial^2 S}{\partial x^2} &\approx \frac{1}{2} \frac{S_{j+1,i} - 2S_{j,i} + S_{j-1,i}}{\Delta x^2} + \frac{1}{2} \frac{S_{j+1,i+1} - 2S_{j,i+1} + S_{j-1,i+1}}{\Delta x^2} \\ S &\approx \frac{S_{j,i} + S_{j,i+1}}{2}. \end{split}$$

Using these approximations we can rewrite equation A.6.32 as

$$\frac{S_{j,i+1} - S_{j,i}}{\Delta t} + (K\theta + KX_t) \frac{S_{j+1,i} - S_{j-1,i} + S_{j+1,i+1} - S_{j-1,i+1}}{4\Delta x} + \frac{1}{2} \frac{S_{j+1,i} - 2S_{j,i} + S_{j-1,i} + S_{j+1,i+1} - 2S_{j,i+1} + S_{j-1,i+1}}{2\Delta x^2} \sigma^2 = e^{X_t} \frac{S_{j,i} + S_{j,i+1}}{2}.$$
(A.6.33)

After some simple algebra calculations equation A.6.33 becomes

$$A_i S_{j+1,i+1} + B_i S_{j,i+1} + C_i S_{j-1,i+1} = -A_i S_{j+1,i} D_i S_{j,i} - C_i S_{j-1,i},$$

that can be written in matrix form as

$$M_2 S_{i+1} = M_1 S_i$$

where $S_i = [S_{1,i}, S_{2,i}, \dots, S_{L-1,i}]^T$,

and

$$M_{2} = \begin{bmatrix} B_{1} & C_{1} & & & \\ A_{2} & B_{2} & C_{2} & & & \\ & A_{3} & B_{3} & C_{3} & & & \\ & & \ddots & \ddots & & \\ & & & & & A_{M-2} & B_{M-2} & C_{M-2} \\ & & & & & & A_{M-1} & B_{M-1} \end{bmatrix}$$

A.7 Price of jump-at-default risk

Consider that the jumps in price occure from time to time: the first jump occurs at time τ_1 , the second occurs τ_2 time units after the first, the third occurs τ_3 units after the second, etc. The variables $\tau_1, \tau_2, \tau_3, \ldots$ are independent exponential random variables, all with the same mean $\frac{1}{\lambda}$. Because the expected time between jumps is $\frac{1}{\lambda}$, the jumps are arriving at an average rate of λ per unit time. The process N_t counts the number of jumps that occur at or before time t, and we say that has intensity λ .

Suppose that under the probability measure \mathbb{P} the Poisson process N_t has intensity $\lambda_t^{\mathbb{P}}$, and thus distribution:

$$P(N_t = k) = \frac{(\lambda^{\mathbb{P}} t)^k}{k!} e^{-\lambda^{\mathbb{P}} t}$$
(A.7.34)

We have that the compensated process $M_t := N_t - \lambda_t^{\mathbb{P}}$ is a martingale; if the intensity is time varying, the process $N_t - \int_0^t \lambda_s^{\mathbb{P}}$ is a \mathbb{P} -martingale. Just as we can use Girsanov's Theorem to change the measure so that a Brownian motion with drift becomes a Brownian motion without drift, and in terms of this change of measure we define the market price of risk, we can change the measure for a Poisson process and define the related jump price of risk. For a Poisson process, the change of measure affects the intensity. We want to change to a measure \mathbb{Q} such that N_t $(0 \le t \le T)$ is a Poisson process with intensity $\lambda_t^{\mathbb{Q}}$, that means that

$$P(N_t = k) = \frac{(\lambda^{\mathbb{Q}} t)^k}{k!} e^{-\lambda^{\mathbb{Q}} t}$$
(A.7.35)

and the process $N_t - \int_0^t \lambda_s^{\mathbb{Q}}$ is a \mathbb{Q} -martingale. Let us fix the time T > 0. The Radon-Nykodim derivative that allows the change of measure is the function

$$Z(T) = e^{(\lambda^{\mathbb{P}} - \lambda^{\mathbb{Q}})T} \left(\frac{\lambda^{\mathbb{Q}}}{\lambda^{\mathbb{P}}}\right)^{N_T}$$
(A.7.36)

and

$$\mathbb{Q}(A) = \int_A Z(T) d\mathbb{P}$$

In fact it can be proved³⁷ that the process Z(t) satisfies

$$dZ(t) = \frac{\lambda^{\mathbb{Q}} - \lambda^{\mathbb{P}}}{\lambda^{\mathbb{P}}} Z(t-) dM(t),$$

is a martingale under \mathbb{P} and $\mathbb{E}Z(t) = 1$ for all t, that is necessary to be able to change the measure.

The jump price of risk can thus be defined as

$$\Gamma_t^{\mathbb{Q}} = \frac{\lambda_t^{\mathbb{Q}}}{\lambda_t^{\mathbb{P}}} - 1 \quad \text{or} \quad \Gamma_t^{\mathbb{P}} = 1 - \frac{\lambda_t^{\mathbb{Q}}}{\lambda_t^{\mathbb{P}}}.$$
(A.7.37)

 $^{^{37}}$ See Shreve (2008) for technical details.

References

- Berndt, A., Douglas, R., Duffie, D., Ferguson, M. Schranz, M. (2005). Measuring Default Risk Premia from Default Swap Rates and EDFs. Working Paper.
- Berndt, A. and Obreja, I. (2010). Decomposing European CDS returns. *Review of Finance*.
- Bharath, S.T. and Shumway, T. (2008). Forecasting Default with the Merton Distance to Default Model. *The Review of Financial Studies*, (21), 1339-1369.
- Blanco, R., Brennan, S. and Marsh, I.W. (2005). An empirical analysis of the dynamic relation between investment-grade bonds and credit defualt swaps. *The Journal of Finance*.
- Cecchetti S. and Taboga, M. (2017). Assessing the risks of asset overvaluation: models and challenges. *Bank of Italy Temi di Discussione (Working Papers)*, No 1114 (April 2017).
- Chen, R. and Scott, L. (1993). Maximum likelihood estimation for a multifactor equilibrium model of the term structure of interest rates. *The Journal of Fixed Income*.
- Cheridito, P., Filipovic, D. and Kimmel, R. L. (2007). Market price of risk specifications for affine models: Theory and evidence. *Journal of Financial Economics*.
- Cont, R. and Tankov, P. (2004). Financial Modelling With Jump Processes. Chapman & Hall/CRC Financial MAthematics Series.
- Diaz, A., Groba, J. and Serrano, P. (2013). What drives corporate default risk premia? Evidence from the CDS market. *Journal of International Money and Finance*.
- Driessen, J. (2005). Is Default Event Risk Priced in Corporate Bonds?. *The Review of Financial Studies*.
- Duffee, G. (2002). Term Premia and Interest Rate Forecasts in Affine Models. *Journal of Finance*.
- Duffie, D. and Singleton, K.J. (1999). Modeling Term Structures of Defaultable Bond Yields. *Review of Financial Studies*.

Duffie, D. and Singleton, K.J. (2003). Credit Risk. Princeton Series in Finance.

- European Central Bank (2016). ECB Economic Bulletin, Issue5 2016, Box 2. Available at : https://www.ecb.europa.eu/pub/pdf/ecbu/eb201605.en.pdf.
- Forte, S. and Pena, J.I. (2009). Credit spreads: an empirical analysis on the informational content of stocks, bonds, and CDS. *Journal of Banking & Finance*.
- Gilchrist, S. and Zakrajsek, E. (2012). Credit Spreads and Business Cycle Fluctuations. American Economic Review.
- Hamilton, J. D. (1994). Time Series Analysis. Princeton University Press.
- Hamilton, J. D. and Wu, J.C. (2012). Identification and Estimation of Gaussian Affine Term Structure Models. NBER Working Paper Series.
- Jarrow, L.A., Lando, D. and Yu F. (2005). Default risk and diversification: theory and empirical implications. *Mathematical finance*.
- Kealhofer, S. (2003). Quantifying credit risk I: default prediction. Financial Analyst Journal, (59), 30–44.
- Kelly, B, Lustig, H. and Nieuwerburgh, S.V.(2012). Too-Systemic-To-Fail: What Option Markets Imply About Sector-wide Government Guarantees. Available at : http://ssrn.com/abstract=1762312.
- Korablev, I. and Dwyer, D. (2007). Power and Level Validation of Moody's KMV EDFTM Credit Measures in North America, Europe and Asia. Available at : http://www.moodysanalytics.com.
- Lando, D. (1998). On Cox Processes and Credit Risky Securities. Review of Derivative Research, (2), 99–120.
- Lando, D. (2004). Credit Risk modeling. Princeton Series in Finance.
- Li, J. and Zinna, G. (2014). On Bank Credit Risk: Systemic or Bank-Specific? Evidence from the US and UK. *Journal of Financial and Quantitative Analysis*.
- Longstaff, F. A., Mithal, S. and Neis, E. (2005). Corporate yields spreads: default risk or liquidity? New evidence from the the credit default swap market. *The Journal of Finance*.

- Longstaff, F. A., Pan, J., Pedersen, L.H. and Singleton, K.J. (2011). How Sovereign is Sovereign Credit Risk? American Economic Journal: Macroeconomics 3.
- Merton, R. C. (1974). On the pricing of corporate debt: the risk structure of interest rates. *The Journal of Finance*, (29), 449–470.
- Pan, J. and Singleton, K.J. (2006). Interpreting Recent Changes in the Credit Spreads of Japanese Banks. *Monetary and economic studies (special edition December 2006)*.
- Pan, J. and Singleton, K.J. (2008). Default and recovery implicit in the term structure of sovereign CDS spread. *The Journal of Finance*.
- Schonbucher, P. J. (2003). Credit derivatives pricing models. Wiley finance series.
- Shreve, S. E. (2008). Stochastic Calculus for Finance II. Continuous-Time Models. Springer.
- Vassalou, M. and Xing, Y. (2004). Default risk in equity returns. The Journal of Finance.
- Yu, F. (2002). Modeling Expected Return on Defaultable Bonds. Journal of Fixed Income, 12(2), 2202, 69–81.

RECENTLY PUBLISHED "TEMI" (*)

- N. 1121 *The financial stability dark side of monetary policy*, by Piergiorgio Alessandri, Antonio Maria Conti and Fabrizio Venditti (June 2017).
- N. 1122 *Large time-varying parameter VARs: a non-parametric approach*, by George Kapetanios, Massimiliano Marcellino and Fabrizio Venditti (June 2017).
- N.1123 *Multiple lending, credit lines, and financial contagion*, by Giuseppe Cappelletti and Paolo Emilio Mistrulli (June 2017).
- N. 1124 *Law enforcement and political participation: Italy, 1861-65*, by Antonio Accetturo, Matteo Bugamelli and Andrea Lamorgese (July 2017).
- N. 1125 The consequences of public employment: evidence from Italian municipalities, by Marta Auricchio, Emanuele Ciani, Alberto Dalmazzo and Guido de Blasio (July 2017).
- N. 1126 *The cyclicality of the income elasticity of trade*, by Alessandro Borin, Virginia Di Nino, Michele Mancini and Massimo Sbracia (July 2017).
- N. 1127 Human capital and urban growth in Italy, 1981-2001, by Francesco Giffoni, Matteo Gomellini and Dario Pellegrino (July 2017).
- N. 1128 *The double bind of asymmetric information in over-the-counter markets*, by Taneli Mäkinen and Francesco Palazzo (July 2017).
- N. 1129 *The effects of central bank's verbal guidance: evidence from the ECB*, by Maddalena Galardo and Cinzia Guerrieri (July 2017).
- N. 1130 The Bank of Italy econometric model: an update of the main equations and model elasticities, by Guido Bulligan, Fabio Busetti, Michele Caivano, Pietro Cova, Davide Fantino, Alberto Locarno, Lisa Rodano (July 2017).
- N.1131 Venture capitalists at work: what are the effects on the firms they finance?, by Raffaello Bronzini, Giampaolo Caramellino and Silvia Magri (September 2017).
- N.1132 Low frequency drivers of the real interest rate: a band spectrum regression approach, by Fabio Busetti and Michele Caivano (September 2017).
- N. 1133 *The real effects of relationship lending*, by Ryan Banerjee, Leonardo Gambacorta and Enrico Sette (September 2017).
- N. 1134 *Credit demand and supply: a two-way feedback relation*, by Ugo Albertazzi and Lucia Esposito (September 2017).
- N. 1135 Legislators' behaviour and electoral rules: evidence from an Italian reform, by Giuseppe Albanese, Marika Cioffi and Pietro Tommasino (September 2017).
- N.1136 Macroeconomic effects of non-standard monetary policy measures in the euro area: the role of corporate bond purchases, by Anna Bartocci, Lorenzo Burlon, Alessandro Notarpietro and Massimiliano Pisani (September 2017).
- N. 1137 On secular stagnation and low interest rates: demography matters, by Giuseppe Ferrero, Marco Gross and Stefano Neri (September 2017).
- N. 1138 Tony Atkinson and his legacy, by Rolf Aaberge, François Bourguignon, Andrea Brandolini, Francisco H. G. Ferreira, Janet C. Gornick, John Hills, Markus Jäntti, Stephen P. Jenkins, Eric Marlier, John Micklewright, Brian Nolan, Thomas Piketty, Walter J. Radermacher, Timothy M. Smeeding, Nicholas H. Stern, Joseph Stiglitz and Holly Sutherland (September 2017).
- N. 1139 *Credit misallocation during the European financial crisis*, by Fabiano Schivardi, Enrico Sette and Guido Tabellini (September 2017).
- N. 1140 Natural rates across the Atlantic, by Andrea Gerali and Stefano Neri (September 2017).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- ALBERTAZZI U., G. ERAMO, L. GAMBACORTA and C. SALLEO, Asymmetric information in securitization: an empirical assessment, Journal of Monetary Economics, v. 71, pp. 33-49, TD No. 796 (February 2011).
- ALESSANDRI P. and B. NELSON, *Simple banking: profitability and the yield curve,* Journal of Money, Credit and Banking, v. 47, 1, pp. 143-175, **TD No. 945 (January 2014).**
- ANTONIETTI R., R. BRONZINI and G. CAINELLI, *Inward greenfield FDI and innovation*, Economia e Politica Industriale, v. 42, 1, pp. 93-116, **TD No. 1006 (March 2015).**
- BARONE G. and G. NARCISO, Organized crime and business subsidies: Where does the money go?, Journal of Urban Economics, v. 86, pp. 98-110, **TD No. 916 (June 2013).**
- BRONZINI R., The effects of extensive and intensive margins of FDI on domestic employment: microeconomic evidence from Italy, B.E. Journal of Economic Analysis & Policy, v. 15, 4, pp. 2079-2109, TD No. 769 (July 2010).
- BUGAMELLI M., S. FABIANI and E. SETTE, The age of the dragon: the effect of imports from China on firmlevel prices, Journal of Money, Credit and Banking, v. 47, 6, pp. 1091-1118, TD No. 737 (January 2010).
- BULLIGAN G., M. MARCELLINO and F. VENDITTI, *Forecasting economic activity with targeted predictors,* International Journal of Forecasting, v. 31, 1, pp. 188-206, **TD No. 847 (February 2012).**
- BUSETTI F., On detecting end-of-sample instabilities, in S.J. Koopman, N. Shepard (eds.), Unobserved Components and Time Series Econometrics, Oxford, Oxford University Press, **TD No. 881 (September 2012).**
- CESARONI T., *Procyclicality of credit rating systems: how to manage it,* Journal of Economics and Business, v. 82. pp. 62-83, **TD No. 1034 (October 2015).**
- CIARLONE A., *House price cycles in emerging economies*, Studies in Economics and Finance, v. 32, 1, **TD No. 863 (May 2012).**
- CUCINIELLO V. and F. M. SIGNORETTI, *Large bank, loan rate markup and monetary policy,* International Journal of Central Banking, v. 11, 3, pp. 141-177, **TD No. 987 (November 2014).**
- DE BLASIO G., D. FANTINO and G. PELLEGRINI, *Evaluating the impact of innovation incentives: evidence from an unexpected shortage of funds,* Industrial and Corporate Change, v. 24, 6, pp. 1285-1314, **TD No. 792 (February 2011).**
- DEPALO D., R. GIORDANO and E. PAPAPETROU, *Public-private wage differentials in euro area countries:* evidence from quantile decomposition analysis, Empirical Economics, v. 49, 3, pp. 985-1115, **TD No. 907 (April 2013).**
- DI CESARE A., A. P. STORK and C. DE VRIES, *Risk measures for autocorrelated hedge fund returns*, Journal of Financial Econometrics, v. 13, 4, pp. 868-895, **TD No. 831 (October 2011).**
- FANTINO D., A. MORI and D. SCALISE, Collaboration between firms and universities in Italy: the role of a firm's proximity to top-rated departments, Rivista Italiana degli economisti, v. 1, 2, pp. 219-251, TD No. 884 (October 2012).
- FRATZSCHER M., D. RIMEC, L. SARNOB and G. ZINNA, *The scapegoat theory of exchange rates: the first tests*, Journal of Monetary Economics, v. 70, 1, pp. 1-21, **TD No. 991 (November 2014).**
- NOTARPIETRO A. and S. SIVIERO, *Optimal monetary policy rules and house prices: the role of financial frictions*, Journal of Money, Credit and Banking, v. 47, S1, pp. 383-410, **TD No. 993 (November 2014).**
- RIGGI M. and F. VENDITTI, *The time varying effect of oil price shocks on euro-area exports,* Journal of Economic Dynamics and Control, v. 59, pp. 75-94, **TD No. 1035 (October 2015).**
- TANELI M. and B. OHL, *Information acquisition and learning from prices over the business cycle,* Journal of Economic Theory, 158 B, pp. 585–633, **TD No. 946 (January 2014).**

- ALBANESE G., G. DE BLASIO and P. SESTITO, *My parents taught me. evidence on the family transmission of values,* Journal of Population Economics, v. 29, 2, pp. 571-592, **TD No. 955 (March 2014).**
- ANDINI M. and G. DE BLASIO, *Local development that money cannot buy: Italy's Contratti di Programma,* Journal of Economic Geography, v. 16, 2, pp. 365-393, **TD No. 915 (June 2013).**
- BARONE G. and S. MOCETTI, *Inequality and trust: new evidence from panel data*, Economic Inquiry, v. 54, pp. 794-809, **TD No. 973 (October 2014).**
- BELTRATTI A., B. BORTOLOTTI and M. CACCAVAIO, *Stock market efficiency in China: evidence from the split-share reform*, Quarterly Review of Economics and Finance, v. 60, pp. 125-137, **TD No. 969** (October 2014).
- BOLATTO S. and M. SBRACIA, *Deconstructing the gains from trade: selection of industries vs reallocation of workers*, Review of International Economics, v. 24, 2, pp. 344-363, **TD No. 1037 (November 2015).**
- BOLTON P., X. FREIXAS, L. GAMBACORTA and P. E. MISTRULLI, *Relationship and transaction lending in a crisis*, Review of Financial Studies, v. 29, 10, pp. 2643-2676, **TD No. 917 (July 2013).**
- BONACCORSI DI PATTI E. and E. SETTE, Did the securitization market freeze affect bank lending during the financial crisis? Evidence from a credit register, Journal of Financial Intermediation, v. 25, 1, pp. 54-76, TD No. 848 (February 2012).
- BORIN A. and M. MANCINI, Foreign direct investment and firm performance: an empirical analysis of *Italian firms*, Review of World Economics, v. 152, 4, pp. 705-732, **TD No. 1011 (June 2015).**
- BRAGOLI D., M. RIGON and F. ZANETTI, *Optimal inflation weights in the euro area*, International Journal of Central Banking, v. 12, 2, pp. 357-383, **TD No. 1045 (January 2016).**
- BRANDOLINI A. and E. VIVIANO, *Behind and beyond the (headcount) employment rate,* Journal of the Royal Statistical Society: Series A, v. 179, 3, pp. 657-681, **TD No. 965 (July 2015).**
- BRIPI F., *The role of regulation on entry: evidence from the Italian provinces*, World Bank Economic Review, v. 30, 2, pp. 383-411, **TD No. 932 (September 2013).**
- BRONZINI R. and P. PISELLI, *The impact of R&D subsidies on firm innovation*, Research Policy, v. 45, 2, pp. 442-457, **TD No. 960 (April 2014).**
- BURLON L. and M. VILALTA-BUFI, A new look at technical progress and early retirement, IZA Journal of Labor Policy, v. 5, **TD No. 963 (June 2014).**
- BUSETTI F. and M. CAIVANO, *The trend-cycle decomposition of output and the Phillips Curve: bayesian estimates for Italy and the Euro Area,* Empirical Economics, V. 50, 4, pp. 1565-1587, **TD No. 941** (November 2013).
- CAIVANO M. and A. HARVEY, *Time-series models with an EGB2 conditional distribution*, Journal of Time Series Analysis, v. 35, 6, pp. 558-571, **TD No. 947 (January 2014).**
- CALZA A. and A. ZAGHINI, *Shoe-leather costs in the euro area and the foreign demand for euro banknotes,* International Journal of Central Banking, v. 12, 1, pp. 231-246, **TD No. 1039 (December 2015).**
- CESARONI T. and R. DE SANTIS, *Current account "core-periphery dualism" in the EMU*, The World Economy, v. 39, 10, pp. 1514-1538, **TD No. 996 (December 2014).**
- CIANI E., *Retirement, Pension eligibility and home production,* Labour Economics, v. 38, pp. 106-120, **TD** No. 1056 (March 2016).
- CIARLONE A. and V. MICELI, Escaping financial crises? Macro evidence from sovereign wealth funds' investment behaviour, Emerging Markets Review, v. 27, 2, pp. 169-196, TD No. 972 (October 2014).
- CORNELI F. and E. TARANTINO, *Sovereign debt and reserves with liquidity and productivity crises*, Journal of International Money and Finance, v. 65, pp. 166-194, **TD No. 1012 (June 2015).**
- D'AURIZIO L. and D. DEPALO, An evaluation of the policies on repayment of government's trade debt in *Italy*, Italian Economic Journal, v. 2, 2, pp. 167-196, **TD No. 1061 (April 2016).**
- DE BLASIO G., G. MAGIO and C. MENON, Down and out in Italian towns: measuring the impact of economic downturns on crime, Economics Letters, 146, pp. 99-102, TD No. 925 (July 2013).
- DOTTORI D. and M. MANNA, *Strategy and tactics in public debt management*, Journal of Policy Modeling, v. 38, 1, pp. 1-25, **TD No. 1005 (March 2015).**
- ESPOSITO L., A. NOBILI and T. ROPELE, *The management of interest rate risk during the crisis: evidence from Italian banks*, Journal of Banking & Finance, v. 59, pp. 486-504, **TD No. 933 (September 2013).**

- LIBERATI D., M. MARINUCCI and G. M. TANZI, Science and technology parks in Italy: main features and analysis of their effects on hosted firms, Journal of Technology Transfer, v. 41, 4, pp. 694-729, TD No. 983 (November 2014).
- MARCELLINO M., M. PORQUEDDU and F. VENDITTI, *Short-Term GDP forecasting with a mixed frequency dynamic factor model with stochastic volatility,* Journal of Business & Economic Statistics, v. 34, 1, pp. 118-127, **TD No. 896 (January 2013).**
- RODANO G., N. SERRANO-VELARDE and E. TARANTINO, *Bankruptcy law and bank financing*, Journal of Financial Economics, v. 120, 2, pp. 363-382, **TD No. 1013 (June 2015).**
- ZINNA G., Price pressures on UK real rates: an empirical investigation, Review of Finance, v. 20, 4, pp. 1587-1630, **TD No. 968 (July 2014).**

2017

- ADAMOPOULOU A. and G.M. TANZI, Academic dropout and the great recession, Journal of Human Capital, V. 11, 1, pp. 35–71, **TD No. 970 (October 2014).**
- ALBERTAZZI U., M. BOTTERO and G. SENE, Information externalities in the credit market and the spell of credit rationing, Journal of Financial Intermediation, v. 30, pp. 61–70, TD No. 980 (November 2014).
- ALESSANDRI P. and H. MUMTAZ, *Financial indicators and density forecasts for US output and inflation*, Review of Economic Dynamics, v. 24, pp. 66-78, **TD No. 977 (November 2014).**
- BARBIERI G., C. ROSSETTI and P. SESTITO, *Teacher motivation and student learning*, Politica economica/Journal of Economic Policy, v. 33, 1, pp.59-72, **TD No. 761 (June 2010).**
- BENTIVOGLI C. and M. LITTERIO, Foreign ownership and performance: evidence from a panel of Italian firms, International Journal of the Economics of Business, v. 24, 3, pp. 251-273, **TD No. 1085** (October 2016).
- BRONZINI R. and A. D'IGNAZIO, *Bank internationalisation and firm exports: evidence from matched firmbank data*, Review of International Economics, v. 25, 3, pp. 476-499 TD No. 1055 (March 2016).
- BRUCHE M. and A. SEGURA, *Debt maturity and the liquidity of secondary debt markets*, Journal of Financial Economics, v. 124, 3, pp. 599-613, **TD No. 1049 (January 2016).**
- BURLON L., *Public expenditure distribution, voting, and growth,* Journal of Public Economic Theory,, v. 19, 4, pp. 789–810, **TD No. 961 (April 2014).**
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Macroeconomic effectiveness of non-standard monetary policy and early exit. a model-based evaluation, International Finance, v. 20, 2, pp.155-173, TD No. 1074 (July 2016).
- BUSETTI F., *Quantile aggregation of density forecasts,* Oxford Bulletin of Economics and Statistics, v. 79, 4, pp. 495-512, **TD No. 979 (November 2014).**
- CESARONI T. and S. IEZZI, *The predictive content of business survey indicators: evidence from SIGE,* Journal of Business Cycle Research, v.13, 1, pp 75–104, **TD No. 1031 (October 2015).**
- CONTI P., D. MARELLA and A. NERI, Statistical matching and uncertainty analysis in combining household income and expenditure data, Statistical Methods & Applications, v. 26, 3, pp 485–505, TD No. 1018 (July 2015).
- D'AMURI F. and J. MARCUCCI, *The predictive power of google searches in forecasting unemployment,* International Journal of Forecasting, v. 33, 4, pp. 801-816, **TD No. 891 (November 2012).**
- DE BLASIO G. and S. POY, *The impact of local minimum wages on employment: evidence from Italy in the* 1950s, Journal of Regional Science, v. 57, 1, pp. 48-74, **TD No. 953 (March 2014).**
- DEL GIOVANE P., A. NOBILI and F. M. SIGNORETTI, Assessing the sources of credit supply tightening: was the sovereign debt crisis different from Lehman?, International Journal of Central Banking, v. 13, 2, pp. 197-234, TD No. 942 (November 2013).
- DELLE MONACHE D. and I. PETRELLA, Adaptive models and heavy tails with an application to inflation forecasting, International Journal of Forecasting, v. 33, 2, pp. 482-501, TD No. 1052 (March 2016).
- DEL PRETE S., M. PAGNINI, P. ROSSI and V. VACCA, Lending organization and credit supply during the 2008–2009 crisis, Economic Notes, v. 46, 2, pp. 207–236, TD No. 1108 (April 2017).
- LOBERTO M. and C. PERRICONE, *Does trend inflation make a difference?*, Economic Modelling, v. 61, pp. 351–375, **TD No. 1033 (October 2015).**

- MANCINI A.L., C. MONFARDINI and S. PASQUA, *Is a good example the best sermon? Children's imitation of parental reading*, Review of Economics of the Household, v. 15, 3, pp 965–993, **D No. 958** (April 2014).
- MEEKS R., B. NELSON and P. ALESSANDRI, *Shadow banks and macroeconomic instability*, Journal of Money, Credit and Banking, v. 49, 7, pp. 1483–1516, **TD No. 939 (November 2013).**
- MICUCCI G. and P. ROSSI, *Debt restructuring and the role of banks' organizational structure and lending technologies*, Journal of Financial Services Research, v. 51, 3, pp 339–361, **TD No. 763 (June 2010).**
- MOCETTI S., M. PAGNINI and E. SETTE, *Information technology and banking organization*, Journal of Journal of Financial Services Research, v. 51, pp. 313-338, **TD No. 752 (March 2010)**.
- MOCETTI S. and E. VIVIANO, *Looking behind mortgage delinquencies*, Journal of Banking & Finance, v. 75, pp. 53-63, **TD No. 999 (January 2015).**
- NOBILI A. and F. ZOLLINO, A structural model for the housing and credit market in Italy, Journal of Housing Economics, v. 36, pp. 73-87, **TD No. 887 (October 2012).**
- PALAZZO F., Search costs and the severity of adverse selection, Research in Economics, v. 71, 1, pp. 171-197, **TD No. 1073 (July 2016).**
- PATACCHINI E. and E. RAINONE, Social ties and the demand for financial services, Journal of Financial Services Research, v. 52, 1–2, pp 35–88, TD No. 1115 (June 2017).
- PATACCHINI E., E. RAINONE and Y. ZENOU, *Heterogeneous peer effects in education*, Journal of Economic Behavior & Organization, v. 134, pp. 190–227, **TD No. 1048 (January 2016).**
- SBRANA G., A. SILVESTRINI and F. VENDITTI, *Short-term inflation forecasting: the M.E.T.A. approach,* International Journal of Forecasting, v. 33, 4, pp. 1065-1081, **TD No. 1016 (June 2015).**
- SEGURA A. and J. SUAREZ, *How excessive is banks' maturity transformation?*, Review of Financial Studies, v. 30, 10, pp. 3538–3580, **TD No. 1065 (April 2016).**
- VACCA V., An unexpected crisis? Looking at pricing effectiveness of heterogeneous banks, Economic Notes, v. 46, 2, pp. 171–206, TD No. 814 (July 2011).
- VERGARA CAFFARELI F., One-way flow networks with decreasing returns to linking, Dynamic Games and Applications, v. 7, 2, pp. 323-345, **TD No. 734 (November 2009).**
- ZAGHINI A., A Tale of fragmentation: corporate funding in the euro-area bond market, International Review of Financial Analysis, v. 49, pp. 59-68, **TD No. 1104 (February 2017).**

FORTHCOMING

- ADAMOPOULOU A. and E. KAYA, Young Adults living with their parents and the influence of peers, Oxford Bulletin of Economics and Statistics, **TD No. 1038 (November 2015).**
- ALBANESE G., G. DE BLASIO and P. SESTITO, *Trust, risk and time preferences: evidence from survey data,* International Review of Economics, **TD No. 911 (April 2013).**
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis,* Journal of the European Economic Association, **TD No. 909 (April 2013).**
- CASIRAGHI M., E. GAIOTTI, L. RODANO and A. SECCHI, A "Reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, Journal of International Money and Finance, **TD No. 1077 (July 2016).**
- D'AMURI F., Monitoring and disincentives in containing paid sick leave, Labour Economics, TD No. 787 (January 2011).
- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, **TD No. 877 (September 2012).**
- GIACOMELLI S. and C. MENON, *Does weak contract enforcement affect firm size? Evidence from the neighbour's court,* Journal of Economic Geography, **TD No. 898 (January 2013).**
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, **TD No. 1025 (July 2015).**
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, **TD No. 871 (July 2012).**
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, TD No. 1100 (February 2017).